	Official Recor- Recording requested By SOUTHEASTERN LINCOLN COUNTY HABITAT
APN	Lincoln County - NV Leslie Boucher - Recorder
APN	Fee: Page 1 of 48 RPTT: Recorded By. AE Book- 256 P age- 0090
APN	0125919
Federal Fish & Wildlife, Permit - Southeasto	(a Lincoln County Habitat Cancernate Class
Title of Doc	ument
Affirmation St	atement
I, the undersigned hereby affirm that the attache submitted for recording does not contain the social secur 239B.030)	ity number of any person or persons. (Per NRS
I, the undersigned hereby affirm that the attache submitted for recording does contain the social security r	ed document, including any exhibits, hereby umber of a person or persons as required by
law:(State specific law)	
Signature Clerk Title Lisa C. Lloyd	
5-11-10	
Date	
	•
Grantees address and mail tax statement:	

DOC # 0135919

09 03 AM

Record

05/11/2010

Page 2 of 48

DEPARTMENT OF THE INTERIOR U.S. FISH AND WILDLIFE SERVICE

FEDERAL FISH AND WILDLIFE PERMIT

1. PERMITTEE

COUNTY OF LINCOLN, NEVADA

-1 MAIN STREET PIOCHE, NV 89043 U.S.A.

	2. AUTHORITY-STATUTES	
	16 USC 1539(a)	
	16 USC 1533(d)	
	16 USC 703-712	
	\wedge	
	REGULATIONS	
	50 CFR 17.22	
	50 CFR 17.32	
	50 CFR 21.23 & 21.27	
	50 CFR 13	
i	\	\
	3. NUMBER	
	TE09163A-0	\.
-	4. RENEWABLE	5. MAY COPY
	YES	YES
	NO	NO
	6. EFFECTIVE	7. EXPIRES
	05/05/2010	05/05/2040
		7. 7.

3-201

(1/97)

8. NAME AND TITLE OF PRINCIPAL OFFICER (##1 is a business) PAUL MATHEWS

CHAIRPERSON OF THE BOARD OF COMMISSIONERS

9. TYPE OF PERMIT

NATIVE ENDANGERED & THREATENED SP. HABITAT CONSERVATION PLAN - E & T WILDLIFE; MIGRATORY BIRDS

10. LOCATION WHERE AUTHORIZED ACTIVITY MAY BE CONDUCTED

Southeastern Lincoln County, Nevada, as described in Section 2.1 of the Southeastern Lincoln County Habitat Conservation Plan

11. CONDITIONS AND AUTHORIZATIONS:

- A. GENERAL CONDITIONS SET OUT IN SUBPART D OF 50 CFR 13, AND SPECIFIC CONDITIONS CONTAINED IN FEDERAL REGULATIONS CITED IN BLOCK #2 ABOVE, ARE HEREBY MADE A PART OF THIS PERMIT. ALL ACTIVITIES AUTHORIZED HEREIN MUST BE CARRIED OUT IN ACCORD WITH AND FOR THE PURPOSES DESCRIBED IN THE APPLICATION SUBMITTED. CONTINUED VALIDITY, OR RENEWAL, OF THIS PERMIT IS SUBJECT TO COMPLETE AND TIMELY COMPLIANCE WITH ALL APPLICABLE CONDITIONS, INCLUDING THE FILING OF ALL REQUIRED INFORMATION AND REPORTS.
- B. THE VALIDITY OF THIS PERMIT IS ALSO CONDITIONED UPON STRICT OBSERVANCE OF ALL APPLICABLE FOREIGN, STATE, LOCAL OR OTHER FEDERAL LAW.
- C. VALID FOR USE BY PERMITTEE NAMED ABOVE.
- D. Further conditions of authorization are contained in the attached Special Terms and Conditions.

ADDITIONAL CONDITIONS AND AUTHORIZATIONS ALSO APPLY

2. REPORTING REQUIREMENTS

ACTING DEPUTY REGIONAL DIRECTOR

DATE 05/05/2010

SOUTHEASTERN LINCOLN COUNTY HABITAT CONSERVATION PLAN SPECIAL TERMS AND CONDITIONS FOR PERMIT TE-09163A-0 FOR LINCOLN COUNTY

- E. All sections of Title 50 Code of Federal Regulations, parts 13, 17.22, and 17.32, and any future amendments thereto, are conditions of this Permit. The current version of these regulations is provided in Attachment 1.
- F. The authorization granted by this Permit is subject to compliance with, and implementation of, the Southeastern Lincoln County Habitat Conservation Plan (HCP) dated January 2010, which is hereby incorporated into this Permit. This Permit and the HCP are binding upon the Permittee, and any authorized officer, employee, contractor, or agent conducting the Covered Activities as described in the HCP.
- G. The Permittee and designated agents are authorized under the Federal Endangered Species Act of 1973, as amended (Act), to incidentally take (harass, harm, injure, and/or kill) the following covered animal species listed under the Act within the 30,673.5-acre Permit Area as described and specified in section 5 of the HCP, to the extent that take of these species would otherwise be prohibited under section 9 of the Act, and its implementing regulations, or pursuant to a rule promulgated under section 4(d) of the Act:

Desert tortoise (Gopherus agassizii) (Mojave population) - Threatened Southwestern willow flycatcher (Empidonax traillii extimus) - Endangered

Take must occur incidental to otherwise lawful activities associated with the development activities as described in the HCP, and as conditioned herein.

- H. This section 10(a)(1)(B) Permit shall constitute a Special Purpose Permit under 50 Code of Federal Regulations (CFR) section 21.27, for the take of Covered Species listed under the Act and which are also listed under the Migratory Bird Treaty Act (MBTA; 16 U.S.C. §§703-712), in the amount and/or number specified in the HCP, subject to the terms and conditions specified in the section 10(a)(1)(B) Permit. Any such take will not be in violation of the MBTA. The MBTA Special Purpose Permit will extend to Covered Species listed under the Act and also under the MBTA after the Effective Date of the section 10(a)(1)(B) Permit. Take, as defined by 50 CFR 10.12, associated with habitat loss for bird species on the list of Covered Species is avoided or minimized within the Planning Area of the HCP by conservation measures provided in section 6 of the HCP and the associated Biological Opinion, and the Special Terms and Conditions of this section 10(a)(1)(B) Permit.
- I. A copy of this Permit must be on file with the Permittee. Please refer to the Permit number in all correspondence and reports concerning Permit activities. Any questions you may have about this Permit should be directed to the Assistant Field Supervisor,

Nevada Fish and Wildlife Office, 4701 North Torrey Pines Drive, Las Vegas, Nevada 89130, 702-515-5230.

- J. The Permittee shall ensure the conservation measures proposed in the Southeastern Lincoln County HCP and repeated in the Biological Opinion on the Issuance of Section 10(a)(1)(B) Incidental Take Permits for the Southeastern Lincoln County HCP are implemented. Reporting requirements shall be completed in accordance with section 8.2 of the HCP.
- K. The Permittee or designated agent shall contact the Nevada Fish and Wildlife Office in Las Vegas at 702-515-5230 within one business day to report any findings of dead, injured, or sick endangered or threatened wildlife species. Written notification should be provided to the Nevada Fish and Wildlife Office in Las Vegas (4701 North Torrey Pines Drive, Las Vegas, Nevada, 89130, attention Assistant Field Supervisor) and the Division of Law Enforcement (208) 378-5333 within five business days and shall include the date, time, and location of the specimen and any other pertinent information.
- All desert tortoise habitat to be disturbed by land development activities as defined in the L. HCP shall be surveyed for desert tortoises prior to commencement of construction projects. As provided in Section 6.6.1.1 of the HCP, Roadway Improvements and Maintenance Activities, Lincoln County will be responsible for conducting pre-activity surveys and clearance in desert tortoise habitat where Service authorized biologists are reasonably certain that the species exists. Pre-disturbance surveys will not be required within sites where a qualified biologist has determined that desert tortoises are not expected to occur. Desert tortoises discovered within proposed maintenance or construction areas will be avoided to the extent possible. If avoidance is not possible, tortoises will be moved out of harm's way by utilizing previously approved Service procedures. For all land development activities or new road construction, a qualified desert tortoise biologist authorized by the Service shall survey for desert tortoises and their burrows using techniques providing 100 percent coverage of the project area. The authorized biologist shall comply with Service approved handling protocols (Desert Tortoise Council 1994, revised 1999, or most current version), and must be familiar with the Special Terms and Conditions of this Permit. Those seeking to be an authorized biologist for the project shall complete the Qualifications Form (see attached Appendix C of the HCP or Appendix A of the Biological Opinion) and submit it to the Service for review and approval as appropriate. Allow 30 days for Service review and response.
- M. As provided in Section 6.6.1.1 of the HCP, prior to initiation of construction activities, a desert tortoise education program shall be presented to all personnel who will be on-site, including but not limited to Permittee employees, contractors, contractors' employees, supervisors, inspectors, and subcontractors who are under the control of the County. The program will advise the workers on the biology and ecology of the desert tortoise, the definition of "take", the threats to the desert tortoise, the mitigation measures, the responsibility for avoiding effects that may result in "take", and the potential fines if "take" occurs that is inconsistent with the authority to be granted by the incidental take permit and the HCP. The Service authorized biologist will provide training for the

County trainers to deliver the appropriate information as a solid reference and overview of the regulatory framework and species issues. Prior to maintenance or construction activities in suitable habitat areas, the County trainers would deliver this information to the employees and subcontractors of the County. In situations where sensitive resources are at risk and specific mitigation measures are required, the County trainer will deliver the worker education program at the site. All personnel who receive the training shall sign a statement indicating that they have completed the education program and understand fully its provisions.

- N. Except as provided in Section 6.6.1.1 of the HCP, collection and possession of desert tortoises detected during pre-construction surveys or construction and maintenance activities is prohibited. Tortoises that are detected during pre-construction surveys for land development activities will either be transferred to the Desert Tortoise Conservation Center located south of Las Vegas, Nevada, or will be relocated as directed by the Service. For utility and road projects initiated by Lincoln County, desert tortoises that are detected during pre-construction surveys or construction and maintenance activities will be moved out of harm's way to adjacent land administered by the Bureau of Land Management (BLM).
- O. All surface disturbance activities associated with land development shall occur inside the temporary or permanent tortoise-proof fencing or other appropriate barrier, as applicable. An Authorized Biologist shall be on-site during installation of the fence or barrier to ensure that no tortoises are harmed. The Authorized Biologist shall thoroughly examine the proposed fence line and burrows for the presence of tortoises and eggs no more than three days before construction commences. The developer will decide the specific type of barrier to be used for each development project, with assistance from the Service if requested. Fencing or other appropriate barriers must be installed and maintained so that they prevent tortoises from gaining access to construction sites. Desert tortoise clearance surveys will be conducted within the construction site after the fence or barrier has been installed.
- P. All potential desert tortoise burrows located within desert tortoise clearance areas shall be excavated by hand by an Authorized Biologist, tortoises and eggs removed, and burrows collapsed or blocked to prevent occupation by desert tortoises or other wildlife species. These activities shall be conducted in accordance with Service approved protocol (Desert Tortoise Council 1994, revised 1999, or most current version). If the Service or Desert Tortoise Council releases a revised protocol before initiation of project activities, the revised protocol shall be implemented for the project area.
- Q. Desert tortoises shall be treated in a manner to ensure that they do not overheat, exhibit signs of overheating (e.g., gaping, foaming at the mouth, etc.), or are placed in a situation where they cannot maintain surface and core temperatures necessary for their well-being. Desert tortoises shall be kept shaded at all times until it is safe to release them. No desert tortoise shall be captured, moved, transported, released, or purposefully caused to leave its burrow for whatever reason when the ambient air temperature is above 95° F (35° C). Ambient air temperature shall be measured in the shade, protected from wind, at a height

of 2 inches (5 centimeters) above the ground surface. No desert tortoise shall be captured if the ambient air temperature is anticipated to exceed 95° F before handling and relocation can be completed. If the ambient air temperature exceeds 95° F during handling or processing, desert tortoises shall be kept shaded in an environment that does not exceed 95° F. Special precautions shall be taken to ensure that desert tortoises are not harmed as a result of their capture and movement during extreme temperatures (i.e., air temperatures below 55° F or above 95° F). Under such adverse conditions, tortoises captured will be monitored continually by an Authorized Biologist or approved monitor(s) until the tortoise exhibits normal behavior. If a desert tortoise shows signs of heat stress, procedures will be implemented as identified in the Service approved protocol (Desert Tortoise Council 1994, revised 1999, or most current revision if available).

- R. For land development activities or in new construction areas, all project personnel shall be instructed that their activities shall be confined to locations within areas previously cleared of tortoises. Travel routes within the project area should be established, cleared of tortoises, and clearly marked prior to construction in any particular area.
- S. Trash and food items will be disposed of properly in predator-proof containers with resealing lids. During construction activities, trash containers will be emptied and waste will be removed frequently from the project area. Upon completion of individual structures or activities in an area, all construction refuse, including but not limited to, broken equipment parts, wrapping material, cords, cables, wire, rope, strapping, twine, buckets, metal or plastic containers, and boxes, will be removed from the site and disposed of properly.
- T. Domestic dogs and cats shall be prohibited from active construction sites in the Development Areas (Lincoln County Land Act (LCLA) area, Alamo Industrial Park and Community Expansion area, or other development areas within the Permit Area).
- U. As described in Section 6.3.1.1.3 of the HCP, possession of desert tortoises or other nonnative tortoise species is prohibited within the Development Areas (LCLA area, Alamo Industrial Park and Community Expansion area, or other residential, commercial, and industrial development areas within the Permit Area).
- V. Prior to full build-out of the LCLA development area and the Alamo Industrial Park and Community Expansion area, desert tortoise-proof fencing or other appropriate barriers shall be constructed around the entire boundary of these developments to control access to developments by desert tortoises and to control access to adjacent tortoise habitat by humans and domestic animals.
- W. The Permittee shall ensure that all desert tortoise-proof fencing or other appropriate barriers are maintained and repaired periodically to ensure fencing remains effective in preventing tortoise access to development areas.
- X. Prior to initiation of construction activities, a southwestern willow flycatcher education program shall be presented to all personnel who will be on-site, including but not limited

to Permittee employees, contractors, contractors' employees, supervisors, inspectors, and subcontractors. The program will advise the workers on the biology and ecology of the flycatcher, the definition of "take", the threats to the flycatcher, the mitigation measures, the responsibility for avoiding effects that may result in "take", and the potential fines if "take" occurs that is inconsistent with the authority to be granted by the incidental take permit and the HCP. This program also will include methods for avoiding flycatchers where possible in those areas and under those circumstances described in Section 6.7.1.2 of the HCP (Pre-Disturbance Surveys and Clearances). The Service authorized biologist will provide training to Lincoln County staff, who will then deliver the appropriate information as a solid reference and overview of the regulatory framework and species issues. Prior to maintenance or construction activities in sensitive habitat areas, the Lincoln County trainers would deliver this information to the employees and subcontractors of Lincoln County. In situations where sensitive resources are at risk, the Lincoln County trainers will deliver the worker education program at the site. All personnel who receive this training shall sign a statement indicating that they have completed the education program and understand fully its provisions.

- Y. Prior to proposed construction activities during the nesting season and in the vicinity of historic nesting sites, surveys for southwestern willow flycatchers shall be conducted at historic nesting sites as described in the HCP. Surveys shall be conducted by County staff members who have participated in training developed by a Service authorized southwestern willow flycatcher biologist to verify the presence or absence of active nests in the proposed work area.
- Z. If at all possible, the Permittee shall avoid conducting construction activities in active nesting areas until all flycatchers disperse from the area. If nesting flycatchers are known to occupy a proposed construction site, and the proposed action does not require an immediate or urgent response, the Permittee shall notify the Service and determine if options are available for avoiding the area or minimizing the potential for take to occur.
- AA. Each year, the Permittees (Lincoln County, City of Caliente, and UPRR) will develop an Annual Work Plan that will identify goals and objectives for the upcoming year, describe the various tasks to be accomplished, identify who will conduct the work, and outline a schedule of events and budgets for the year. An Implementation and Monitoring Committee (IMC) will be appointed by the Board of Lincoln County Commissioners (BLCC) to oversee implementation of the HCP and develop the Annual Work Plan. At a minimum, membership will consist of a representative each from Lincoln County, City of Caliente, and UPRR. In addition, a representative each from the Service, BLM, and Lincoln County Conservation District will serve on the committee on an ad hoc advisory capacity. IMC membership may be expanded to include other entities and consultants upon approval by the BLCC and concurrence from the Service. IMC members will review, comment, and make recommendations to the BLCC regarding prioritized Conservation Measures (minimization and mitigation measures) and budget proposals.

The IMC will present the proposed Annual Work Plan to the BLCC for preliminary approval consistent with the County's annual budgeting process. The BLCC will then provide the proposed Annual Work Plan to the Service for review and comment. Within 30 days, the Service will provide a concurrence review of the Annual Work Plan to the BLCC to ensure that (1) all avoidance, minimization, and mitigation measures are commensurate with the level of effect in any one year, and (2) the Annual Work Plan is adapted as appropriate to address priority conservation needs of the Covered Species and to respond to monitoring results. The Service must concur with the proposed Annual Work Plan prior to final approval by the BLCC. Upon receipt of concurrence from the Service, the proposed Annual Work Plan will be presented to the BLCC for final approval and implementation. In the event the IMC, BLCC, and/or the Service cannot reach agreement on the proposed Annual Work Plan, the involved parties will work together in good faith to resolve any disputes, and if necessary, follow the terms for dispute resolution as described in Section 14.5 of the Implementing Agreement.

Reference Cited

Desert Tortoise Council. 1994. Guidelines for handling desert tortoises during construction projects. E.L. LaRue, Jr. ed. San Bernardino, California. Revised 1999.



§ 17.21

```
560—69 FR 59177; November 16, 1994,
564—59 FR 60568; November 25, 1994,
565—59 FR 62352; December 25, 1991.
567-59 FR 64623; December 15, 1991.
      -60 FR 61; January 3, 1995.
572—60 FR 3562; January 18, 1995.
575—60 FR 6684; February 3, 1995.
578—60 FR 12486; March 7, 1995.
      -61 FR 10697; March 15, 1996.
584-61 FR 31058; June 19, 1996.
586-61 FR 41023; August 7, 1996.
587—61 FR 43184; August 21, 1996.
589—61 FR 52384; October 7, 1996.
590—61 FR 53688; October 10, 1996.
591-61 FR 53107; October 10, 1996.
592-61 FR 53123; October 10, 1998.
      -61 FR 53130; October 10, 1996.
      -61 FR 53137; October 10, 1996.
595—61 FR 53152; October 10, 1996.
596—61 FR 54358; October 18, 1996.
599—61 FR 67497; December 23, 1996.
600-62 FR 689; January 6, 1997.
601-62 FR 1647; January 10, 1997.
603-62 FR 1694; January 13, 1997
606-62 FR 4182; January 29, 1997.
609-62 FR 5551; February 6, 1997.
611-62 FR 14351; March 26, 1997.
      -62 FR 31748; June 11, 1997.
619—62 FR 33037; June 18, 1997.
620—62 FR 33373; June 19, 1997.
623-62 FR 40973; July 31, 1997.
524-62 FR 42702; August 8, 1997.
525-52 FR 54807; October 22, 1997.
627-62 FR 61925; November 20, 1997.
635-63 FR 19849; April 22, 1998
640-63 FR 43115; August 12, 1998.
641—63 FR 44594; August 20, 1998.
643—63 FR 49034; September 14, 1998.
614-63 FR 49021; September 14, 1998.
617-63 FR 63615; October 6, 1998.
648—63 FR 64970; October 13, 1998.
649—63 FR 64956; October 13, 1998.
650—63 FR 64994; October 13, 1998.
      -63 FR 59244; November 3, 1998,
657—64 FR 13120; March 17, 1999.
661—64 FR 28412; May 26, 1999.
662—64 FR 28403; May 26, 1999.
663—64 FR 28392; May 26, 1999.
      -64 FR 48323; September 3, 1999.
      -64 FR 56590; October 20, 1999.
668-64 FR 56596; October 20, 1999.
671-61 FR 63752; November 22, 1999.
672-61 FR 69203; December 10, 1999.
673-64 FR 71687; December 22, 1999.
678—65 FR 3875; Jamary 25, 2000.
679—65 FR 3890; January 25, 2000.
681—65 FR 4162; January 26, 2000.
683-65 FR 5275; February 3, 2000.
684-65 FR 6338; February 9, 2000.
      -85 FR 7764; February 16, 2000.
689-65 FR 14887; March 20, 2000.
690-65 FR 14909; March 20, 2000.
691-65 FR 14897; March 20, 2000.
704 65 FR 62310; October 18, 2000.
708—66 FR 27908; May 21, 2001.
711—66 FR 49567; September 28, 2001.
712-66 FR 51606; October 10, 2001.
```

50 CFR Ch. I (10-1-08 Edition)

720-67 FR 1668; January 14, 2002. 721—67 FR 3125; January 23, 2002. 722—67 FR 5526; February 6, 2002. 723—67 FR 11449; March 14, 2002. 727—67 FR 44382; July 2, 2002. 733—67 FR 68015; November 7, 2002. 736—68 FR 16988; April 8, 2003. 739—68 FR 59344; October 15, 2003. 742 - 69 FR 18506, April 8, 2004.

EDITORIAL NOTE 1: FOR FEDERAL REGISTER citations affecting the table in \$17.12(h), see the listing above.

EDITORIAL NOTE 2: For FEDERAL REGISTER citations affecting \$17.12, see the List of OFR Sections Affected, which appears in the Finding Aids section of the printed volume and on GPO Access.

Subpart C-Endangered Wildlife

§17.21 Prohibitions.

(a) Except as provided in subpart A of this part, or under permits issued pursuant to §17.22 or §17.23, it is unlawful for any person subject to the jurisdic-tion of the United States to commit, to attempt to commit, to solicit another to commit or to cause to be com-mitted, any of the acts described in paragraphs (b) through (f) of this section in regard to any endangered wildlife.

(b) Import or export. It is unlawful to import or to export any endangered wildlife. Any shipment in transit through the United States is an importation and an exportation, whether or not it has entered the country for cus-

toms purposes.

(c) Take. (1) It is unlawful to take endangered wildlife within the United States, within the territorial sea of the United States, or upon the high seas. The high seas shall be all waters seaward of the territorial sea of the United States, except waters officially recognized by the United States as the territorial sea of another country, under international law.

(2) Notwithstanding paragraph (c)(1) of this section, any person may take endangered wildlife in defense of his own life or the lives of others.

(3) Notwithstanding paragraph (c)(1) of this section, any employee or agent of the Service, any other Federal land management agency, the National Marine Fisheries Service, or a State conservation agency, who is designated by his agency for such purposes, may,

\$ 17.21

when acting in the course of his official duties, take endangered wildlife without a permit if such action is necessary to:

- (1) Aid a sick, injured or orphaned specimen; or
- (ii) Dispose of a dead specimen; or (iii) Salvage a dead specimen which

may be useful for scientific study; or

- (iv) Remove specimens which constitute a demonstrable but nonimmediate threat to human safety, provided that the taking is done in a humane manner; the taking may involve killing or injuring only if it has not been reasonably possible to eliminate such threat by live-capturing and releasing the specimen unharmed, in a remote area.
- (4) Any taking under paragraphs (c)(2) and (3) of this section must be reported in writing to the U.S. Fish and Wildlife Service, Office of Law Enforcement, 4401 North Fairfax Drive, LE-3000, Arlington, VA 22203, within five days. The specimen may only be retained, disposed of, or salvaged under directions from the Office of Law Enforcement.
- (5) Notwithstanding paragraph (c)(1) of this section, any qualified employee or agent of a State Conservation Agency which is a party to a Cooperative Agreement with the Service in accordance with section 6(c) of the Act, who is designated by his agency for such purposes, may, when acting in the course of his official duties take those endangered species which are covered by an approved cooperative agreement for conservation programs in accordance with the Cooperative Agreement, provided that such taking is not reasonably anticipated to result in:
- (i) The death or permanent disabling of the specimen;
- (ii) The removal of the specimen from the State where the taking occurred:
- (iii) The introduction of the specimen so taken, or of any progeny derived from such a specimen, into an area beyond the historical range of the species: or
- (iv) The holding of the specimen in captivity for a period of more than 45 consecutive days.
- (6) Notwithstanding paragraph (c)(1) of this section, any person acting under

a valid migratory bird rehabilitation permit issued pursuant to §21.31 of this subchapter may take endangered migratory birds without an endangered species permit if such action is necessary to aid a sick, injured, or orphaned endangered migratory bird, provided the permittee:

(i) Notifies the issuing Migratory Bird Permit Office immediately upon receipt of such bird (contact information for your issuing office is listed on your permit and on the Internet at

http://offices.fws.gov); and

(ii) Disposes of or transfers such birds, or their parts or feathers, as directed by the Migratory Bird Permit Office.

- (7) Notwithstanding paragraph (c)(1) of this section, persons exempt from the permit requirements of §21.12(c) and (d) of this subchapter may take sick and injured endangered migratory birds without an endangered species permit in performing the activities authorized under §21.12(c) and (d).
- (d) Possession and other acts with unlawfully taken wildlife. (1) It is unlawful to possess, sell, deliver, carry, transport, or ship, by any means whatsoever, any endangered wildlife which was taken in violation of paragraph (c) of this section.

Example. A person captures a whooping crane in Texas and gives it to a second person, who puts it in a closed van and drives thirty miles, to another location in Texas. The second person then gives the whooping crane to a third person, who is apprehended with the bird in his possession. All three have violated the law—the first by illegally taking the whooping crane; the second by transporting an illegally taken whooping crane; and the third by possessing an illegally taken whooping owne.

- (2) Notwithstanding paragraph (d)(1) of this section, Federal and State law enforcement officers may possess, deliver, carry, transport or ship any endangered wildlife taken in violation of the Act as necessary in performing their official duties.
- (3) Notwithstanding paragraph (d)(1) of this section, any person acting under a valid migratory bird rehabilitation permit issued pursuant to §21.31 of this subchapter may possess and transport endangered migratory birds without an endangered species permit when such

action is necessary to aid a sick, injured, or orphaned endangered migratory bird, provided the permittee:

(i) Notifies the issuing Migratory Bird Permit Office immediately upon receipt of such bird (contact information for your issuing office is listed on your permit and on the Internet at http://offices.fws.gov); and

(ii) Disposes of or transfers such birds, or their parts or feathers, as directed by the Migratory Bird Permit

Office.

(4) Notwithstanding paragraph (d)(1) of this section, persons exempt from the permit requirements of §21.12(c) and (d) of this subchapter may possess and transport sick and injured endangered migratory bird species without an endangered species permit in performing the activities authorized under §21.12(e) and (d).

(e) Interstate or foreign commerce. It is unlawful to deliver, receive, carry transport, or ship in interstate or foreign commerce, by any means whatsoever, and in the course of a commercial activity, any endangered wildlife.

(f) Sale or offer for sale. (1) It is unlawful to sell or to offer for sale in interstate or foreign commerce any endan-

gered wildlife.

- (2) An advertisement for the sale of endangered wildlife which carries a warning to the effect that no sale may be consummated until a permit has been obtained from the U.S. Fish and Wildlife Service shall not be considered an offer for sale within the meaning of this section.
- (g) Captive-bred wildlife. (1) Notwithstanding paragraphs (b), (c), (e) and (f) of this section, any person may take; export or re-import; deliver, receive, carry, transport or ship in interstate or foreign commerce, in the course of a commercial activity; or sell or offer for sale in interstate or foreign commerce any endangered wildlife that is bred in captivity in the United States provided either that the wildlife is of a taxon listed in paragraph (g)(6) of this sec-tion, or that the following conditions are met:
- (i) The wildlife is of a species having a natural geographic distribution not including any part of the United States, or the wildlife is of a species that the Director has determined to be

eligible in accordance with paragraph (g)(5) of this section:

(ii) The purpose of such activity is to enhance the propagation or survival of the affected species;

(iii) Such activity does not involve interstate or foreign commerce, in the course of a commercial activity, with respect to non-living wildlife;

(iv) Each specimen of wildlife to be re-imported is uniquely identified by a band, tattoo or other means that was reported in writing to an official of the Service at a port of export prior to export from the United States and

(v) Any person subject to the jurisdiction of the United States who engages in any of the activities authorized by this paragraph does so in accordance with paragraphs (g) (2), (3) and (4) of this section, and with all other applicable regulations in this Subchapter B.

(2) Any person subject to the jurisdiction of the United States seeking to engage in any of the activities authorized by this paragraph must first register with the Service (Office of Management Authority, U.S. Fish and Wildlife Service, 4401 N. Fairfax Drive, Arlington, Virginia 22203). Requests for registration must be submitted on an official application form (Form 3-200-41) provided by the Service, and must include the following information:

(i) The types of wildlife sought to be covered by the registration, identified by common and scientific name to the taxonomic level of family, genus or

(ii) A description of the applicant's experience in maintaining and propagating the types of wildlife sought to be covered by the registration, and when appropriate, in conducting re-search directly related to maintaining and propagating such wildlife;

(iii) Photograph(s) or other evidence clearly depicting the facilities where such wildlife will be maintained; and

(iv) a copy of the applicant's license or registration, if any, under the animal welfare regulations of the U.S. Department of Agriculture (9 CFR part 2).

(3) Upon receiving a complete application, the Director will decide whether or not the registration will be approved. In making this decision, the Director will consider, in addition to

§ 17.21

the general criteria in §13.21(b) of this subchapter, whether the expertise, facilities or other resources available to the applicant appear adequate to enhance the propagation or survival of the affected wildlife, Public education activities may not be the sole basis to justify issuance of a registration or to otherwise establish eligibility for the exception granted in paragraph (g)(1) of this section. Each person so registered must maintain accurate written records of activities conducted under the registration, and allow reasonable access to Service agents for inspection purposes as set forth in §§ 13.46 and 13.47. Each person registered must submit to the Director an individual written annual report of activities, including all births, deaths and transfers of any type.

(4) Any person subject to the jurisdiction of the United States seeking to export or conduct foreign commerce in captive-bred endangered wildlife that will not remain under the care of that person must first obtain approval by providing written evidence to satisfy the Director that the proposed recipient of the wildlife has expertise, facilities or other resources adequate to enhance the propagation or survival of such wildlife and that the proposed recipient will use such wildlife for purposes of enhancing the propagation or survival of the affected species.

(5)(1) The Director will use the following criteria to determine if wildlife of any species having a natural geographic distribution that includes any part of the United States is eligible for the provisions of this paragraph:

(A) Whether there is a low demand for taking of the species from wild populations, either because of the success of captive breeding or because of other reasons, and

(B) Whether the wild populations of the species are effectively protected from unauthorized taking as a result of the inaccessibility of their habitat to humans or as a result of the effectiveness of law enforcement.

(ii) The Director will follow the procedures set forth in the Act and in the regulations thereunder with respect to petitions and notification of the public and governors of affected States when

determining the eligibility of species for purposes of this paragraph.

(iii) In accordance with the criteria in paragraph (g)(5)(i) of this section, the Director has determined the following species to be eligible for the provisions of this paragraph:

Laysan duck (Anas laysanensis).

(6) Any person subject to the jurisdiction of the United States seeking to engage in any of the activities authorized by paragraph (g)(1) of this section may do so without first registering with the Service with respect to the bar-tailed pheasant (Syrmaticus humiae), Elliot's pheasant (S. ellioti), Mikado pheasant (S. mikado), brown eared pheasant (Crossoptilon mantchuricum), eared pheasant (C. crossoptilon), cheer pheasant (Catreus wallichit), Edward's pheasant (Lophura edwardsi), Swinhoe's pheasant (L. swinholl), Chinese monal (Lophophorus lhuysii), and Palawan peacock pheasant (Polyplectron emphanum); parakeets of the species Neophema pulchella and N. splendida, the Laysan duck (Anas laysanensis); the whitewinged wood duck (Cairina scutulata); and the inter-subspecific crossed or 'generic' tiger (Panthera ligris) (t e., specimens not identified or identifiable as members of the Bengal, Sumatran, Siberian or Indochinese subspecies (Panthera ligris tigris, P.t. sumatrae, P.t. altaica and P.t. corbetti, respectively) provided:

(i) The purpose of such activity is to enhance the propagation or survival of the affected exempted species:

(ii) Such activity does not involve interstate or foreign commerce, in the course of a commercial activity, with respect to non-living wildlife;

(iii) Each specimen to be re-imported is uniquely identified by a band, tattoo or other means that was reported in writing to an official of the Service at a port of export prior to export of the specimen from the United States:

(iv) No specimens of the taxa in this paragraph (g)(6) of this section that were taken from the wild may be imported for breeding purposes absent a definitive showing that the need for new bloodlines can only be met by wild specimens, that suitable foreign-bred, captive individuals are unavailable, and that wild populations can sustain

limited taking, and an import permit is issued under §17.22;

- (v) Any permanent exports of such specimens meet the requirements of paragraph (g)(4) of this section; and
- (vi) Each person claiming the benefit of the exception in paragraph (g)(1) of this section must maintain accurate written records of activities, including births, deaths and transfers of specimens, and make those records accessible to Service agents for inspection at reasonable hours as set forth in §§ 13.46 and 13.47.
- (h) U.S. captive-bred scimitar-horned oryx, addax, and dama gazelle. Notwithstanding paragraphs (b), (c), (e), and (f) of this section, any person subject to the jurisdiction of the United States may take; export or re-import; deliver, receive, carry, transport or ship in interstate or foreign commerce, in the course of a commercial activity; or sell or offer for sale in interstate or foreign commerce live wildlife, including embryos and gametes, and sport-hunted trophies of scimitar-horned cryx (Organ dammah), addax (Addax nasomaculatus), and dama gazelle (Gazella dama) provided:
- (1) The purpose of such activity is associated with the management or transfer of live wildlife, including embryos and gametes, or sport hunting in a manner that contributes to increasing or sustaining captive numbers or to potential reintroduction to range countries;
- (2) The specimen was captive-bred, in accordance with §17.3, within the United States;
- (3) All live specimens of that species held by the captive-breeding operation are managed in a manner that prevents hybridization of the species or subspecies.
- (4) All live specimens of that species held by the captive-breeding operation are managed in a manner that maintains genetic diversity.
- (5) Any export of or foreign commerce in a specimen meets the requirements of paragraph (g)(4) of this section, as well as parts 13, 14, and 23 of this chapter:
- (6) Each specimen to be re-imported is uniquely identified by a tattoo or other means that is reported on the

documentation required under paragraph (h)(5) of this section; and

- (7) Each person claiming the benefit of the exception of this paragraph (h) must maintain accurate written records of activities, including births, deaths, and transfers of specimens, and make those records accessible to Service officials for inspection at reasonable hours set forth in §§ 13.46 and 13.47 of this chapter.
- (8) The sport-hunted trophy consists of raw or tanned parts, such as bones, hair, head, hide, hooves, horns, meat, skull, rug, taxidermied head, shoulder, or full body mount, of a specimen that was taken by the hunter during a sport hunt for personal use. It does not include articles made from a trophy, such as worked, manufactured, or handicraft items for use as clothing, curios, ornamentation, jewelry, or other utilitarian items for commercial purposes.

[40 FR 44415, Sept. 26, 1975, as amended at 40 FR 63490, Nov. 18, 1975; 41 FR 19226, May 11, 1976; 44 FR 31580, May 31, 1979; 44 FR 54007, Sept. 17, 1979; 58 FR 68325, Dec. 27, 1993; 63 FR 48640, Sept. 11, 1996; 68 FR 2919, Jan. 22, 2003; 68 FR 61136, Oct. 27, 2003; 79 FR 52318, Sept. 2, 2005; 73 FR 47733, Aug. 14, 2008]

§17.22 Permits for scientific purposes, enhancement of propagation or sur-vival, or for incidental taking.

Upon receipt of a complete application, the Director may issue a permit authorizing any activity otherwise prohibited by §17.21, in accordance with the Issuance criteria of this section, for scientific purposes, for enhancing the propagation or survival, or for the incidental taking of endangered wildlife. Such permits may authorize a single transaction, a series of transactions, or a number of activities over a specific period of time. (See §17.32 for permits for threatened species.) The Director shall publish notice in the FEDERAL REGISTER of each application for a permit that is made under this section. Each notice shall invite the submission from interested parties, within 30 days after the date of the notice, of written data, views, or arguments with respect to the application. The 30-day period may be waived by the Director in an emergency situation where the life or health of an endangered animal is

threatened and no reasonable alternative is available to the applicant. Notice of any such waiver shall be published in the FEDERAL REGISTER within 10 days following issuance of the permit.

- (a)(1) Application requirements for permits for scientific purposes or for the enhancement of propagation or survival. A person wishing to get a permit for an activity prohibited by §17.21 submits an application for activities under this paragraph. The Service provides Form 3-200 for the application to which all of the following must be attained:
- (i) The common and scientific names of the species sought to the covered by the permit, as well as the number, age, and sex of such species, and the activity sought to be authorized (such as taking, exporting, selling in interstate commerce);
- (ii) A statement as to whether, at the time of application, the wildlife sought to be covered by the permit (A) is still in the wild, (B) has already been removed from the wild, or (C) was born in captivity:
- (iii) A resume of the applicant's attempts to obtain the wildlife sought to be covered by the permit in a manner which would not cause the death or removal from the wild of such wildlife;
- (iv) If the wildlife sought to be covered by the permit has already been removed from the wild, the country and place where such removal occurred; if the wildlife sought to be covered by the permit was born in captivity, the country and place where such wildlife was born:
- (v) A complete description and address of the institution or other facility where the wildlife sought to be covered by the permit will be used, displayed, or maintained;
- (vi) If the applicant seeks to have live wildlife covered by the permit, a complete description, including photographs or diagrams, of the facilities to house and/or care for the wildlife and a resume of the experience of those person who will be caring for the wildlife;
- (vii) A full statement of the reasons why the applicant is justified in obtaining a permit including the details of the activities sought to be authorized by the permit;

(viii) If the application is for the purpose of enhancement of propagation, a statement of the applicant's willingness to participate in a cooperative breeding program and to maintain or contribute data to a studbook:

(2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (a)(1) of this section, the Director will decide whether or not a permit should be issued. In making this decision, the Director shall consider, in addition to the general criteria in §13.21(b) of this subchapter, the following factors:

(1) Whether the purpose for which the permit is required is adequate to justify removing from the wild or otherwise changing the status of the wildlife sought to be covered by the permit;

(ii) The probable direct and indirect effect which issuing the permit would have on the wild populations of the wildlife sought to be covered by the permit;

(iii) Whether the permit, if issued, would in any way, directly or indirectly, conflict with any known program intended to enhance the survival probabilities of the population from which the wildlife sought to be covered by the permit was or would be removed:

(iv) Whether the purpose for which the permit is required would be likely to reduce the threat of extinction facing the species of wildlife sought to be covered by the permit;

(v) The opinions or views of scientists or other persons or organizations having expertise concerning the wildlife or other matters germane to the application: and

(vi) Whether the expertise, facilities, or other resources available to the applicant appear adequate to successfully accomplish the objectives stated in the application.

(3) Permit conditions. In addition to the general conditions set forth in part 13 of this subchapter, every permit issued under this paragraph shall be subject to the special condition that the escape of living wildlife covered by the permit shall be immediately reported to the Service office designated in the permit.

(4) Duration of permits. The duration of permits issued under this paragraph shall be designated on the face of the permit.

(b)(1) Application requirements for permits for incidental taking. A person wishing to get a permit for an activity prohibited by \$17.21(c) submits an application for activities under this paragraph. The Service provides Form 3-200 for the application to which all of the following must be attached:

(i) A complete description of the activity sought to be authorized:

(ii) The common and scientific names of the species sought to be covered by the permit, as well as the number, age, and sex of such species, if known;

(iii) A conservation plan that specifies:

(A) The impact that will likely result from such taking;

(B) What steps the applicant will take to monitor, minimize, and mitigate such impacts, the funding that will be available to implement such steps, and the procedures to be used to deal with unforeseen circumstances;

(C) What alternative actions to such taking the applicant considered and the reasons why such alternatives are not proposed to be utilized; and

(D) Such other measures that the Director may require as being necessary or appropriate for purposes of the plan;

- (2) Issuance criteria. (i) Upon receiving an application completed in accordance with paragraph (b)(1) of this section, the Director will decide whether or not a permit should be issued. The Director shall consider the general issuance criteria in §13.21(b) of this subchapter, except for §13.21(b)(4), and shall issue the permit if he or she finds that:
 - (A) The taking will be incidental;
- (B) The applicant will, to the maximum extent practicable, minimize and mitigate the impacts of such takings;
- (C) The applicant will ensure that adequate funding for the conservation plan and procedures to deal with unforeseen circumstances will be pro-
- (D) The taking will not appreciably reduce the likelihood of the survival and recovery of the species in the wild;
- (E) The measures, if any, required under paragraph (b)(1)(iii)(D) of this section will be met; and

(F) He or she has received such other assurances as he or she may require that the plan will be implemented.

(ii) In making his or her decision, the Director shall also consider the anticipated duration and geographic scope of the applicant's planned activities, including the amount of listed species habitat that is involved and the degree to which listed species and their habitats are affected.

(3) Permit conditions. In addition to the general conditions set forth in part 13 of this subchapter, every permit issued under this paragraph shall contain such terms and conditions as the Director deems necessary or appropriate to carry out the purposes of the permit and the conservation plan including, but not limited to, monitoring and reporting requirements deemed necessary for determining whether such terms and conditions are being complied with. The Director shall rely upon existing reporting requirements to the maximum extent practicable.

(4) Duration of permits. The duration of permits issued under this paragraphshall be sufficient to provide adequate assurances to the permittee to commit funding necessary for the activities authorized by the permit, including conservation activities and land use restrictions. In determining the duration of a permit, the Director shall consider the duration of the planned activities, as well as the possible positive and negative effects associated with permits of the proposed duration on listed species, including the extent to which the conservation plan will enhance the habitat of listed species and increase the longterm survivability of such species.

(5) Assurances provided to permittee in case of changed or unforeseen circumstances. The assurances in this paragraph (b)(5) apply only to incidental take permits issued in accordance with paragraph (b)(2) of this section where the conservation plan is being properly implemented, and apply only with respect to species adequately covered by the conservation plan. These assurances cannot be provided to Federal agencies. This rule does not apply to incidental take permits issued prior to March 25, 1998. The assurances provided in incidental take permits issued prior to March 25, 1998 remain in

§ 17.22

U.S. Fish and Wildlife Serv., Interior

effect, and those permits will not be revised as a result of this rulemaking.

(i) Changed circumstances provided for in the plan. If additional conservation and mitigation measures are deemed necessary to respond to changed circumstances and were provided for in the plan's operating conservation program, the permittee will implement the measures specified in the plan.

(ii) Changed circumstances not provided for in the plan. If additional conservation and mitigation measures are deemed necessary to respond changed circumstances and such measures were not provided for in the plan's operating conservation program, the Director will not require any conservation and mitigation measures in addition to those provided for in the plan without the consent of the permittee, provided the plan is being properly implemented.

(iii) Unforeseen circumstances. (A) In negotiating unforeseen circumstances, the Director will not require the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or other natural resources beyond the level otherwise agreed upon for the specles covered by the conservation plan without the consent of the permittee.

(B) If additional conservation and mitigation measures are deemed necessary to respond to unforeseen circumstances, the Director may require additional measures of the permittee where the conservation plan is being properly implemented, but only if such measures are limited to modifications within conserved habitat areas, if any, or to the conservation plan's operating conservation program for the affected species, and maintain the original terms of the conservation plan to the maximum extent possible, Additional conservation and mitigation measures will not involve the commitment of additional land, water or financial compensation or additional restrictions on the use of land, water, or other natural resources otherwise available for development or use under the original terms of the conservation plan without the consent of the permittee.

(C) The Director will have the burden of demonstrating that unforeseen circumstances exist, using the best scientific and commercial data available. These findings must be clearly documented and based upon reliable technical information regarding the status and habitat requirements of the affected species. The Director will consider, but not be limited to, the following factors:

(1) Size of the current range of the affected species;

(2) Percentage of range adversely affected by the conservation plan;

(3) Percentage of range conserved by the conservation plan;

(4) Ecological significance of that portion of the range affected by the conservation plan;

(5) Level of knowledge about the affeeted species and the degree of specificity of the species' conservation program under the conservation plan; and

(6) Whether failure to adopt additional conservation measures would appreciably reduce the likelihood of survival and recovery of the affected specles in the wild.

(6) Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local, or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included in a conservation plan.

(7) Discontinuance of permit activity. Notwithstanding the provisions of §13.26 of this subchapter, a permittee under this paragraph (b) remains responsible for any outstanding minimization and mitigation measures required under the terms of the permit for take that occurs prior to surrender of the permit and such minimization and mitigation measures as may be required pursuant to the termination provisions of an implementing agreement, habitat conservation plan, or permit even after surrendering the permit to the Service pursuant to §13.28 of this subchapter. The permit shall be deemed canceled only upon a determination by the Service that such minimization and mitigation measures have been implemented. Upon surrender of the permit, no further take shall be authorized under the terms of the surrendered permit.

- (8) Criteria for revocation. A permit issued under paragraph (b) of this section may not be revoked for any reason except those set forth in \$13.26(a)(1) through (4) of this subchapter or unless continuation of the permitted activity would be inconsistent with the criterion set forth in 16 U.S.C. 1539(a)(2)(B)(iv) and the inconsistency has not been remedied.
- (c)(1) Application requirements for permits for the enhancement of survival through Safe Harbor Agreements. The applicant must submit an application for a permit under this paragraph (c) to the appropriate Regional Director, U.S. Fish and Wildlife Service, for the Region where the applicant resides or where the proposed activity is to occur (for appropriate addresses, see 50 CFR 10.22), if the applicant wishes to engage in any activity prohibited by §17.21. The applicant must submit an official Service application form (3-200.54) that includes the following information:
- The common and scientific names of the listed species for which the applicant requests incidental take authorization;
- (ii) A description of how incidental take of the listed species pursuant to the Safe Harbor Agreement is likely to occur, both as a result of management activities and as a result of the return to baseline; and
- (iii) A Safe Harbor Agreement that complies with the requirements of the Safe Harbor policy available from the Service.
- (2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (c)(1) of this section, the Director will decide whether or not to issue a permit. The Director shall consider the general issuance criteria in §13.21(b) of this subchapter, except for §13.21(b)(4), and may issue the permit if he or she finds:
- The take will be incidental to an otherwise lawful activity and will be in accordance with the terms of the Safe Harbor Agreement;
- (ii) The implementation of the terms of the Safe Harbor Agreement is reasonably expected to provide a net conservation benefit to the affected listed species by contributing to the recovery of listed spacies included in the permit, and the Safe Harbor Agreement other-

wise complies with the Safe Harbor policy available from the Service;

- (iii) The probable direct and indirect effects of any authorized take will not appreciably reduce the likelihood of survival and recovery in the wild of any listed species;
- (iv) Implementation of the terms of the Safe Harbor Agreement is consistent with applicable Federal, State, and Tribal laws and regulations;
- (v) Implementation of the terms of the Safe Harbor Agreement will not be in conflict with any ongoing conservation or recovery programs for listed species covered by the permit; and
- (vi) The applicant has shown capability for and commitment to implementing all of the terms of the Safe Harbor Agreement.
- (3) Permit conditions. In addition to any applicable general permit conditions set forth in part 13 of this subchapter, every permit issued under this paragraph (c) is subject to the following special conditions:
- A requirement for the participating property owner to notify the Service of any transfer of lands subject to a Safe Harbor Agreement;
- (ii) When appropriate, a requirement for the permittee to give the Service reasonable advance notice (generally at least 30 days) of when he or she expects to incidentally take any listed species covered under the permit. Such notification will provide the Service with an epportunity to relocate affected individuals of the species, if possible and appropriate; and
- (iii) Any additional requirements or conditions the Director deems necessary or appropriate to carry out the purposes of the permit and the Safe Harbor Agreement.
- (4) Permit effective date. Permits issued under this paragraph (c) become effective the day of issuance for species covered by the Safe Harbor Agreement.
- (5) Assurances provided to permittee. (1) The assurances in paragraph (c)(5) (ii) of this section (c)(5) apply only to Safe Harbor permits issued in accordance with paragraph (c)(2) of this section where the Safe Harbor Agreement is being properly implemented, and apply only with respect to species covered by the Agreement and permit. These assurances cannot be provided to Federal

agencies. The assurances provided in this section apply only to Safe Harbor permits issued after July 19, 1999.

(ii) The Director and the permittee may agree to revise or modify the management measures set forth in a Safe Harbor Agreement if the Director determines that such revisions or modifications do not change the Director's prior determination that the Safe Harbor Agreement is reasonably expected to provide a net conservation benefit to the listed species. However, the Director may not require additional or different management activities to be undertaken by a permittee without the consent of the permittee.

(6) Additional actions. Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included

in a Safe Harbor Agreement.

(7) Criteria for revocation. The Director may not revoke a permit issued under paragraph (c) of this section except as provided in this paragraph. The Director may revoke a permit for any reason set forth in §13.28(a)(1) through (4) of this subchapter. The Director may revoke a permit if continuation of the permitted activity would either appreciably reduce the likelihood of survival and recovery in the wild of any listed species or directly or indirectly alter designated critical habitat such that it appreciably diminishes the value of that critical habitat for both the survival and recovery of a listed species. Before revoking a permit for either of the latter two reasons, the Director, with the consent of the permittee, will pursue all appropriate options to avoid permit revocation. These options may include, but are not limited to: extending or modifying the existing permit, capturing and relocating the species, compensating the landowner to forgo the activity, purchasing an easement or fee simple interest in the property, or arranging for a thirdparty acquisition of an interest in the property.

(8) Duration of permits. The duration of permits issued under this paragraph (c) must be sufficient to provide a net conservation benefit to species covered in the enhancement of survival permit. In determining the duration of a permit, the Director will consider the duration of the planned activities, as well as the positive and negative effects associated with permits of the proposed duration on covered species, including the extent to which the conservation activities included in the Safe Harbor Agreement will enhance the survival and contribute to the recovery of listed species included in the permit.

(d)(1) Application requirements for permits for the enhancement of survival through Candidate Conservation Agreements with Assurances. The applicant must submit an application for a permit under this paragraph (d) to the appropriate Regional Director, U.S. Fish and Wildlife Service, for the Region where the applicant resides or where the proposed activity is to occur (for appropriate addresses, see 50 CFR 10.22). When a species covered by a Candidate Conservation Agreement with Assurances is listed as endangered and the applicant wishes to engage in activities identified in the Agreement and otherwise prohibited by §17.31, the applicant must apply for an enhancement of survival permit for species covered by the Agreement. The permit will become valid if and when covered proposed, candidate or other unlisted species is listed as an endangered species. The applicant must submit an official Service application form (3-200.54) that includes the following information:

(i) The common and scientific names of the species for which the applicant requests incidental take authorization;

(ii) A description of the land use or water management activity for which the applicant requests incidental take authorization; and

(iii) A Candidate Conservation Agreement that complies with the requirements of the Candidate Conservation Agreement with Assurances policy available from the Service.

(2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (d)(1) of this section, the Director will decide whether or not to issue a permit. The Director shall consider the general issuance criteria in §13.21(b) of this subchapter, except

for §13.21(b)(4), and may issue the permit if he or she finds:

- (i) The take will be incidental to an otherwise lawful activity and will be in accordance with the terms of the Candidate Conservation Agreement;
- (ii) The Candidate Conservation Agreement complies with the requirements of the Candidate Conservation Agreement with Assurances policy available from the Service:
- (iii) The probable direct and indirect effects of any authorized take will not appreciably reduce the likelihood of survival and recovery in the wild of any species;
- (iv) Implementation of the terms of the Candidate Conservation Agreement is consistent with applicable Federal, State, and Tribal laws and regulations;
- (v) Implementation of the terms of the Candidate Conservation Agreement will not be in conflict with any ongoing conservation programs for species covered by the permit; and
- (vi) The applicant has shown capability for and commitment to implementing all of the terms of the Candidate Conservation Agreement.
- (3) Permit conditions. In addition to any applicable general permit conditions set forth in part 13 of this subchapter, every permit issued under this paragraph (4) is subject to the following special conditions:
- (i) A requirement for the property owner to notify the Service of any transfer of lands subject to a Candidate Conservation Agreement;
- (ii) When appropriate, a requirement for the permittee to give the Service reasonable advance notice (generally at least 30 days) of when he or she expects to incidentally take any listed species covered under the permit. Such notification will provide the Service with an opportunity to relocate affected individuals of the species, if possible and appropriate; and
- (iii) Any additional requirements or conditions the Director deems necessary or appropriate to carry out the purposes of the permit and the Candidate Conservation Agreement.
- (4) Permit effective date. Permits issued under this paragraph (d) become effective for a species covered by a Candidate Conservation Agreement on

the effective date of a final rule that lists a covered species as endangered.

- (5) Assurances provided to permittee in case of changed or unforeseen circumstances. The assurances in this paragraph (d)(5) apply only to permits issued in accordance with paragraph (d)(2) where the Candidate Conservation with Assurances Agreement is being properly implemented, and apply only with respect to species adequately covered by the Candidate Conservation with Assurances Agreement. These assurances cannot be provided to Federal agencies.
- (i) Changed circumstances provided for in the Agreement. If the Director determines that additional conservation measures are necessary to respond to changed circumstances and these measures were set forth in the Agreement, the permittee will implement the measures specified in the Agreement.
- (ii) Changed circumstances not provided for in the Agreement. If the Director determines that additional conservation measures not provided for in the Agreement are necessary to respond to changed circumstances, the Director will not require any conservation measures in addition to those provided for in the Agreement without the consent of the permittee, provided the Agreement is being properly implemented.
- (iii) Unforeseen circumstances. (A) In negotiating unforeseen circumstances, the Director will not require the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or other natural resources beyond the level otherwise agreed upon for the species covered by the Agreement without the consent of the permittee.
- (B) If the Director determines additional conservation measures are necessary to respond to unforeseen circumstances, the Director may require additional measures of the permittee where the Agreement is being properly implemented, but only if such measures maintain the original terms of the Agreement to the maximum extent Additional possible. conservation measures will not involve the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or

other natural resources otherwise available for development or use under the original terms of the Agreement without the consent of the permittee.

- (C) The Director will have the burden of demonstrating that unforeseen circumstances exist, using the best scientific and commercial data available. These findings must be clearly documented and based upon reliable technical information regarding the status and habitat requirements of the affected species. The Director will consider, but not be limited to, the following factors:
- (1) Size of the current range of the affected species;
- (2) Percentage of range adversely affected by the Agreement;
- (3) Percentage of range conserved by the Agreement;
- (4) Ecological significance of that portion of the range affected by the Agreement;
- (5) Level of knowledge about the affected species and the degree of specificity of the species' conservation program under the Agreement; and
- (6) Whether failure to adopt additional conservation measures would appreciably reduce the likelihood of survival and recovery of the affected species in the wild.
- (6) Additional actions. Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included in a Candidate Conservation with As-

surances Agreement.

(7) Criteria for revocation. The Director may not revoke a permit issued under paragraph (d) of this section except as provided in this paragraph. The Director may revoke a permit for any reason set forth in §13.28(a)(1) through (4) of this subchapter. The Director may revoke a permit if continuation of the permitted activity would either appreciably reduce the likelihood of survival and recovery in the wild of any listed species or directly or indirectly alter designated critical habitat such that it appreciably diminishes the value of that critical habitat for both the survival and recovery of a listed species. Before revoking a permit for

either of the latter two reasons, the Director, with the consent of the permittee, will pursue all appropriate options to avoid permit revocation. These options may include, but are not limited to: extending or modifying the existing permit, capturing and relocating the species, compensating the landowner to forgo the activity, purchasing an easement or fee simple interest in the property, or arranging for a third-party acquisition of an interest in the property.

- (8) Duration of the Candidate Conservation Agreement. The duration of a Candidate Conservation Agreement covered by a permit issued under this
 paragraph (d) must be sufficient to enable the Director to determine that the
 benefits of the conservation measures
 in the Agreement, when combined with
 those benefits that would be achieved if
 it is assumed that the conservation
 measures would also be implemented
 on other necessary properties, would
 preclude or remove any need to list the
 species covered by the Agreement.
- (e) Objection to permit issuance. (1) In regard to any notice of a permit application published in the FEDERAL REG-ISTER, any interested party that objects to the issuance of a permit, in whole or in part, may, during the comment period specified in the notice, request notification of the final action to be taken on the application. A separate written request shall be made for each permit application. Such a request shall specify the Service's permit application number and state the reasons why that party believes the applicant does not meet the issuance criteria contained in §§13.21 and 17.22 of this subchapter or other reasons why the permit should not be issued.
- (2) If the Service decides to issue a permit contrary to objections received pursuant to paragraph (c)(1) of this section, then the Service shall, at least ten days prior to issuance of the permit, make reasonable efforts to contact by telephone or other expedient means, any party who has made a request pursuant to paragraph (c)(1) of this section and inform that party of the issuance of the permit. However, the Service may reduce the time period or dispense with such notice if it determines that time is of the essence and

§ 17.23

that delay in issuance of the permit would: (i) Harm the specimen or population involved; or (ii) unduly hinder the actions authorized under the permit.

(3) The Service will notify any party filing an objection and request for notice under paragraph (c)(1) of this section of the final action taken on the application, in writing. If the Service has reduced or dispensed with the notice period referred to in paragraph (c)(2) of this section, it will include its reasons therefore in such written notice.

[50 FR 39687, Sept. 39, 1985, as amended at 63 FR 8871, Feb. 23, 1998; 63 FR 52625, Oct. 1, 1998; 64 FR 32711, June 17, 1999; 64 FR 52676, Sept. 30, 1989; 69 FR 24092, May 3, 2004; 69 FR 29670, May 25, 2004; 69 FR 71731, Dec. 10, 2004]

§ 17.23 Economic hardship permits,

Upon receipt of a complete application, the Director may issue a permit authorizing any activity otherwise prohibited by §17.21, in accordance with the issuance criteria of this section in order to prevent undue economic hardship. The Director shall publish notice in the FEDERAL REGISTER of each application for a permit that is made under this section. Each notice shall invite the submission from interested parties. within 30 days after the date of the notice, of written data, views, or arguments with respect to the application. The 30-day period may be waived by the Director in an emergency situation where the life or health of an endangered animal is threatened and no reasonable alternative is available to the applicant. Notice of any such waiver shall be published in the FEDERAL REG-ISTER within 10 days following issuance of the permit.

(a) Application requirements. Applications for permits under this section must be submitted to the Director by the person allegedly suffering undue economic hardship because his desired activity is prohibited by §17.21. Each application must be submitted on an official application form (Form 3-200) provided by the Service, and must include, as an attachment, all of the information required in §17.22 plus the following additional information:

50 CFR Ch. I (10-1-08 Edition)

- The possible legal, economic or subsistence alternatives to the activity sought to be authorized by the permit;
- (2) A full statement, accompanied by copies of all relevant contracts and correspondence, showing the applicant's involvement with the wildlife sought to be covered by the permit (as well as his involvement with similar wildlife), including, where applicable, that portion of applicant's income derived from the taking of such wildlife, or the subsistence use of such wildlife, during the calendar year immediately preceding either the notice in the FED-ERAL REGISTER of review of the status of the species or of the proposal to list such wildlife as endangered, whichever is earliest:
- (3) Where applicable, proof of a contract or other binding legal obligation which:
- (i) Deals specifically with the wildlife sought to be covered by the permit;
- (ii) Became binding prior to the date when the notice of a review of the status of the species or the notice of proposed rulemaking proposing to list such wildlife as endangered was published in the FEDERAL REGISTER, whichever is earlier; and
- (iii) Will cause monetary loss of a given dollar amount if the permit sought under this section is not granted.
- (b) issuance criteria. Upon receiving an application completed in accordance with paragraph (a) of this section, the Director will decide whether or not a permit should be issued under any of the three categories of economic hardship, as defined in section 10(b)(2) of the Act. In making his decisions, the Director shall consider, in addition to the general criteria in §13.21(b) of this subchapter, the following factors:
- (1) Whether the purpose for which the permit is being requested is adequate to justify removing from the wild or otherwise changing the status of the wildlife sought to be covered by the permit;
- (2) The probable direct and indirect effect which issuing the permit would have on the wild populations of the wildlife sought to be covered by the permit;

§ 17.32

- (3) The economic, legal, subsistence, or other alternatives or relief available to the applicant;
- (4) The amount of evidence that the applicant was in fact party to a contract or other binding legal obligation which;
- (i) Deals specifically with the wildlife sought to be covered by the permit;
- (ii) Became binding prior to the date when the notice of a review of the status of the species or the notice of proposed rulemaking proposing to list such wildlife as endangered was published in the FEDERAL REGISTER, whichever is earlier.
- (5) The severity of economic hardship which the contract or other binding legal obligation referred to in paragraph (b)(4) of this section would cause if the permit were denied;
- (6) Where applicable, the portion of the applicant's income which would be lost if the permit were denied, and the relationship of that portion to the balance of his income;
- (7) Where applicable, the nature and extent of subsistence taking generally by the applicant; and
- (8) The likelihood that applicant can reasonably carry out his desired activity within one year from the date a notice is published in the FEDERAL REG-ISTER to review status of such wildlife, or to list such wildlife as endangered. whichever is earlier.
- (c) Permit conditions. In addition to the general conditions set forth in part 13 of this subchapter, every permit issued under this section shall be subject to the following special conditions:
- (1) In addition to any reporting requirements contained in the permit itself, the permittee shall also submit to the Director a written report of his activities pursuant to the permit. Such report must be postmarked or actually delivered no later than 10 days after completion of the activity.
- (2) The death or escape of all living wildlife covered by the permit shall be immediately reported to the Service's office designated in the permit.
- (d) Duration of permits issued under this section shall be designated on the face of the permit. No permit issued under this section, however, shall be valid for more than one year from the

date a notice is published in the FED-ERAL REGISTER to review status of such wildlife, or to list such wildlife as endangered, whichever is earlier.

(40 FR 44415, Sept. 26, 1975, as amended at 40 FR 53400, Nov. 18, 1975; 40 FR 58307, Dec. 16, 1975; 50 FR 39688, Sept. 30, 1985]

Subpart D—Threatened Wildlife

§17.31 Prohibitions.

- (a) Except as provided in subpart A of this part, or in a permit issued under this subpart, all of the provisions in §17.21 shall apply to threatened wildlife, except § 17.21(c)(5).
- (b) In addition to any other provisions of this part 17, any employee or agent of the Service, of the National Marine Fisheries Service, or of a State conservation agency which is operating a conservation program pursuant to the terms of a Cooperative Agreement with the Service in accordance with section 6(c) of the Act, who is designated by his agency for such purposes, may, when acting in the course of his official duties, take those threatened species of wildlife which are covered by an approved cooperative agreement to carry out conservation programs.
- (c) Whenever a special rule in §§17.40 to 17.48 applies to a threatened species, none of the provisions of paragraphs (a) and (b) of this section will apply. The special rule will contain all the applicable prohibitions and exceptions.

(43 FR 18181, Apr. 28, 1978, as amended at 44 FR 31580, May 31, 1979; 70 FR 10503, Mar. 4,

§17.32 Permits—general.

Upon receipt of a complete application the Director may issue a permit for any activity otherwise prohibited with regard to threatened wildlife. Such permit shall be governed by the provisions of this section unless a special rule applicable to the wildlife, appearing in §§ 17.40 to 17.48, of this part provides otherwise. Permits issued under this section must be for one of the following purposes: Scientific purposes, or the enhancement of propagation or survival, or economic hardship, or zoological exhibition, or educational purposes, or incidental taking, or special purposes consistent with the purposes of the Act. Such permits may authorize a single transaction, a series of transactions, or a number of activities over a specific period of time.

(a)(1) Application requirements for permits for scientific purposes, or the enhancement of propagation or survival, or economic hardship, or zoological exhibition, or educational purposes, or special purposes consistent with the purposes of the Act. A person wishing to get a permit for an activity prohibited by §17.31 submits an application for activities under this paragraph. The Service provides Form 3-200 for the application to which as much of the following information relating to the purpose of the permit must be attached:

(i) The Common and scientific names of the species sought to be covered by the permit, as well as the number, age, and sex of such species, and the activity sought to be authorized (such as taking, exporting, selling in interstate commarce):

(ii) A statement as to whether, at the time of application, the wildlife sought to be covered by the permit (A) is still in the wild, (B) has already been removed from the wild, or (C) was born in captivity;

(iif) A resume of the applicant's attempts to obtain the wildlife sought to be covered by the permit in a manner which would not cause the death or removal from the wild of such wildlife;

- (iv) If the wildlife sought to be covered by the permit has already been removed from the wild, the country and place where such removal occurred; if the wildlife sought to be covered by permit was born in captivity, the country and place where such wildlife was born:
- (v) A complete description and address of the institution or other facility where the wildlife sought to be covered by the permit will be used, displayed, or maintained;
- (vi) If the applicant seeks to have live wildlife covered by the permit, a complete description, including photographs or diagrams, of the facilities to house and/or care for the wildlife and a resume of the experience of those persons who will be caring for the wildlife;

(vii) A full statement of the reasons why the applicant is justified in obtaining a permit including the details of the activities sought to be authorized by the permit;

(viii) If the application is for the purpose of enhancement of propagation, a statement of the applicant's willingness to participate in a cooperative breeding program and to maintain or contribute data to a studbook;

(2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (a)(1) of this section, the Director will decide whether or not a permit should be issued. In making this decision, the Director shall consider, in addition to the general criteria in §13.21(b) of this subchapter, the following factors:

(i) Whether the purpose for which the permit is required is adequate to justify removing from the wild or otherwise changing the status of the wildlife sought to be covered by the permit:

(ii) The probable direct and indirect effect which issuing the permit would have on the wild populations of the wildlife sought to be covered by the permit;

(iii) Whether the permit, if issued, would in any way, directly or indirectly, conflict with any known program intended to enhance the survival probabilities of the population from which the widdlife sought to be covered by the permit was or would be removed;

(iv) Whether the purpose for which the permit is required would be likely to reduce the threat of extinction facing the species of wildlife sought to be covered by the permit:

(v) The opinions or views of scientists or other persons or organizations having expertise concerning the wildlife or other matters germane to the application: and

(vi) Whether the expertise, facilities, or other resources available to the applicant appear adequate to successfully accomplish the objectives stated in the application.

(3) Permit conditions. In addition to the general conditions set forth in part 13 of this subchapter, every permit issued under this paragraph shall be subject to the special condition that the escape of living wildlife covered by

§21.23

any other band, clip, dye, or other method of marking is prohibited unless specifically authorized in the permit.

- (2) All traps or nots used to capture migratory birds for banding or marking purposes shall have attached thereto a tag or label clearly showing the name and address of the permittee and his permit number, or the area in which such traps or nets are located must be posted with notice of banding operations posters (Form 3-1155, available upon request from the Bird Banding Laboratory, Office of Migratory Bird Management, U.S. Fish and Wildlife Service, Laurel, Md. 20708) which shall bear the name and address of the permittee and the number of his permit.
- (3) The holder of a banding or marking permit may salvage, for the purpose of donating to a public scientific or educational institution, birds killed or found dead as a result of the permittee's normal banding operations, and casualties from other causes. All dead birds salvaged under authority of a migratory bird banding or marking permit must be donated and transferred to a public scientific or educational institution at least every 6 months or within 60 days of the time such permit expires or is revoked, unless the permittee has been issued a special permit authorizing possession for a longer period of time.
- (4) Permittees must keep accurate records of their operations and file reports as set forth in the North American Bird Banding Manual, or supplements thereto, in accordance with instructions contained therein.
- (d) Term of permit. A banding or marking permit issued or renewed under this part expires on the date designated on the face of the permit unless amended or revoked, but the term of the permit shall not exceed three (3) years from the date of issuance or re-

(39 FR 1178, Jan. 4, 1974, as amended at 54 FR 38151, Sept. 14, 1989]

§ 21.23 Scientific collecting permits.

(a) Permit requirement. A scientific collecting permit is required before any person may take, transport, or possess migratory birds, their parts,

nests, or eggs for scientific research or educational purposes.

- (b) Application procedures. Submit applications for scientific permits to the appropriate Regional Director (Attention: Migratory bird permit office). You can find addresses for the Regional Directors in 50 CFR 2.2. Each application must contain the general information and certification required in §13.12(a) of this subchapter, and the following additional information:
- (1) Species and numbers of migratory birds or their parts, nests, or eggs to be taken or acquired when it is possible to determine same in advance;
- (2) Location or locations where such scientific collecting is proposed;
- (3) Statement of the purpose and justification for granting such a permit, including an outline of any research project involved:
- (4) Name and address of the public. scientific, or educational institution to which all specimens ultimately will be donated; and
- (5) If a State permit is required by State law, a statement as to whether or not the applicant possesses such State permit, giving its number and expiration date.
- (c) Additional permit conditions. In addition to the general conditions set forth in part 13 of this subchapter B, scientific collecting permits shall be subject to the following conditions:
- (1) All specimens taken and possessed under authority of a scientific collecting permit must be donated and transferred to the public scientific, or educational institution designated in the permit application within 60 days following the date such permit expires or is revoked, unless the permittee has been issued a special purpose permit (See §21.27) authorizing possession for a longer period of time.
- (2) Unless otherwise provided on the permit, all migratory game birds taken pursuant to a scientific collecting permit during the open hunting season for such birds must be in conformance with part 20 of this subchapter;
- (3) Unless specifically stated on the permit, a scientific collecting permit does not authorize the taking of live migratory birds from the wild.

§21.24

50 CFR Ch. I (10-1-09 Edition)

(4) In addition to any reporting requirement set forth in the permit, a report of the scientific collecting activities conducted under authority of such permit shall be submitted to the issuing officer on or before January 10 of each calendar year following the year of issue unless a different date is stated in the permit.

(d) Term of permit. A scientific collecting permit issued or renewed under this part expires on the date designated on the face of the permit unless amended or revoked, but the term of the permit shall not exceed three (3) years from the date of issuance or renewal.

[39 FR 1178, Jan. 4, 1974, as amended at 54 FR 38151, Sept. 14, 1989; 63 FR 52637, Oct. 1, 1998]

§21.24 Taxidermist permits.

(a) Permit requirement. A taxidermist permit is required before any person may perform taxidermy services on migratory birds or their parts, nests, or eggs for any person other than himself.

- (b) Application procedures. Submit application for taxidermist permits to the appropriate Regional Director (Attention: Migratory bird permit office). You can find addresses for the Regional Directors in 50 CFR 2.2. Each application must contain the general information and certification required in § 13.12(a) of this subchapter, and the following additional information:
- The address of premises where taxidermist services will be provided;
- (2) A statement of the applicant's qualifications and experience as a taxidermist; and
- (3) If a State permit is required by State law, a statement as to whether or not the applicant possesses such State permit, giving its number and expiration date.
- (c) Permit authorizations. A permit authorizes a taxidermist to:
- (1) Receive, transport, hold in custody or possession, mount or otherwise prepare, migratory birds, and their parts, nests, or eggs, and return them to another.
- (2) Sell properly marked, captive reared migratory waterfowl which he has lawfully acquired and mounted. Such mounted birds may be placed on consignment for sale and may be possessed by such consignee for the purpose of sale.

- (d) Additional permit conditions. In addition to the general conditions set forth in part 13 of this subchapter B, taxidermist permits shall be subject to the following conditions:
- (1) Permittees must keep accurate records of operations, on a calendar year basis, showing the names and addresses of persons from and to whom migratory birds or their parts, nests, or eggs were received or delivered, the number and species of such, and the dates of receipt and delivery. In addition to the other records required by this paragraph, the permittee must maintain in his files, the original of the completed Form 3-186, Notice of Waterfowl Sale or Transfer, confirming his acquisition of captive reared, properly marked migratory waterfowl from the holder of a current waterfowl sale and disposal permit.

(2) Notwithstanding the provisions of paragraph (c) of this section, the receipt, possession, and storage by a taxidermist of any migratory game birds taken by another by hunting is not authorized unless they are tagged as required by §20.36 of this subchapter. The required tags may be removed during the taxidermy operations but must be retained by the taxidermist with the other records required to be kept and must be reattached to the mounted specimen after mounting. The tag must then remain attached until the mounted specimen is delivered to the owner.

(e) Term of permit. A taxidermist permit issued or renewed under this part expires on the date designated on the face of the permit unless amended or revoked, but the term of the permit will not exceed five (5) years from the date of issuance or renewal.

[39 FR 1178, Jan. 4, 1974, as amended at 54 FR 38151, Sept. 14, 1989; 63 FR 52637, Oct. 1, 1998; 70 FR 18320, Apr. 11, 2005]

§21.25 Waterfowl sale and disposal permits.

(a) Permit requirement. A waterfowl sale and disposal permit is required before any person may lawfully sell, trade, donate, or otherwise dispose of, to another person, any species of captive-reared and properly marked migratory waterfowl or their eggs, except that such a permit is not required for such sales or disposals of captive-

may be required by the conditions of the permit to file with the issuing office an annual report of operation. Annual reports, if required, shall be filed no later than January 31 of the calendar year followng the year for which the report is required. Reports, if required, shall describe permitted activities, numbers and species of migratory birds acquired and disposed of, and shall inventory and describe all migratory birds possessed under the special purpose permit on December 31 of the reporting year.

- (2) Permittees shall make such other reports as may be requested by the issuing officer.
- (3) All live, captive-bred, migratory game birds possessed under authority of a valid special purpose permit shall be physically marked as defined in §21.13(b) of this part.
- (4) No captive-bred migratory game bird may be sold or bartered unless marked in accordance with §21.13(b) of this part.
- (5) No permittee may take, purchase, receive or otherwise acquire, sell, barter, transfer, or otherwise dispose of any captive-bred migratory game bird unless such permittee submits a Service form 3-186A (Migratory Bird Acquisition/Disposition Report), completed in accordance with the instructions on the form, to the issuing office within five (5) days of such transaction.
- (6) No permittee, who is authorized to sell or barter migratory game birds pursuant to a permit issued under this section, may sell or barter such birds to any person unless that person is authorized to purchase and possess such migratory game birds under a permit issued pursuant to this part and part 13, or as permitted by regulations in this part.
- (d) Term of permit. A special purpose permit issued or renewed under this part expires on the date designated on the face of the permit unless amended or revoked, but the term of the permit shall not exceed three (3) years from the date of issuance or renewal.
- [39 FR 1178, Jan. 4, 1974, as amended at 54 FR 38152, Sept. 14, 1989; 63 FR 52637, Oct. 1, 1998]

§21.28 [Reserved]

§21.29 Falconry standards and falconry permitting.

- (a) Background—(1) The legal basis for regulating falconry. The Migratory Bird Treaty Act prohibits any person from taking, possessing, purchasing, bartering, selling, or offering to purchase, barter, or sell, among other things, raptors (birds of prey) listed in §10.13 of this subchapter unless the activities are allowed by Federal permit issued under this part and part 13 of this chapter, or as permitted by regulations in this part.
- (i) This section covers all Falconiformes (vultures, kites, eagles, hawks, caracaras, and falcons) and all Strigiformes (owls) listed in §10.13 of this subchapter ("native" raptors), and applies to any person who possesses one or more wild-caught, captive-bred, or hybrid raptors protected under the MBTA to use in falconry.
- (ii) The Bald and Golden Eagle Protection Act (16 U.S.C. 668-668d, 54 Stat. 250) provides for the taking of golden eagles from the wild to use in falconry. It specifies that the only golden eagles that may be used for falconry are those that would be taken because of depredations on livestock or wildlife (16 U.S.C. 668a).
- (2) "Possession" and short-term handling of a falconry raptor. We do not consider short-term handling, such as letting any other person hold or practice flying a raptor you possess under your permit, to be possession for the purposes of this section if you are present and the person is under your supervision.
- (3) Regulatory year for governing falconry. For determining possession and take of raptors for falconry, a year is any 12-month period for take defined by the State, tribe, or territory.
- (b) Federal approval of State, tribal, and territorial falconry programs—(1) General. (1) A State (including the District of Columbia), tribe, or territory under the jurisdiction of the United States that wishes to allow falconry must establish laws and regulations (hereafter referred to as laws) that meet the standards established in this

50 CFR Ch. I (10-1-09 Edition)

§21.27

(e) What are the OMB information collection requirements of the permit program? OMB has approved the information collection requirements of the permit and assigned clearance number 1018-0099. Federal agencies may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number. We will use the information collection requirements to administer this program and in the issuance and monitoring of these special permits. We will require the information from State wildlife agencies responsible for migratory bird management in order to obtain a special Canada goose permit, and to determine if the applicant meets all the permit issuance criteria, and to protect migratory birds. We estimate the public reporting burden for this collection of information to average 8 hours per response for 45 respondents (States), including the time for reviewing instructions, gathering and maintaining data needed, and completing and reviewing the collection of information. Thus, we estimate the total annual reporting and record-keeping for this collection to be 360 hours. States may send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing the burden, to the Service Information Collection Clearance Officer, Fish and Wildlife Service, ms 224-ARLSQ, 1849 C Street N.W., Washington, DC 20240, or the Office of Management and Budget, Paperwork Reduction Project 1018-0099, Washington, DC 20503.

[64 FR 32774, June 17, 1999]

§ 21.27 Special purpose permits.

Permits may be issued for special purpose activities related to migratory birds, their parts, nests, or eggs, which are otherwise outside the scope of the standard form permits of this part. A special purpose permit for migratory bird related activities not otherwise provided for in this part may be issued to an applicant who submits a written application containing the general information and certification required by part 13 and makes a sufficient showing of benefit to the migratory bird resource, important research reasons,

reasons of human concern for individual birds, or other compelling justification.

(a) Permit requirement. A special purpose permit is required before any person may lawfully take, salvage, otherwise acquire, transport, or possess migratory birds, their parts, nests, or eggs for any purpose not covered by the standard form permits of this part. In addition, a special purpose permit is required before any person may sell, purchase, or barter captive-bred, migratory game birds, other than waterfowl, that are marked in compliance with §21.13(b) of this part.

(b) Application procedures. Submit application for special purpose permits to the appropriate Regional Director (Attention: Migratory bird permit office). You can find addresses for the Regional Directors in 50 CFR 2.2. Each application must contain the general information and certification required in §13.12(a) of this subchapter, and the following additional information:

(1) A detailed statement describing the project or activity which requires issuance of a permit, purpose of such project or activity, and a delineation of the area in which it will be conducted. (Copies of supporting documents, research proposals, and any necessary State permits should accompany the application);

(2) Numbers and species of migratory birds involved where same can reasonably be determined in advance; and

(3) Statement of disposition which will be made of migratory birds involved in the permit activity.

(c) Additional permit conditions. Inaddition to the general conditions set forth in part 13 of this subchapter B, special purpose permits shall be subject to the following conditions:

(1) Permittees shall maintain adequate records describing the conduct of the permitted activity, the numbers and species of migratory birds acquired and disposed of under the permit, and inventorying and identifying all migratory birds held on December 31 of each calendar year. Records shall be maintained at the address listed on the permit; shall be in, or reproducible in English; and shall be available for inspection by Service personnel during regular business hours. A permittee

§ 17.32

the permit shall be immediately reported to the Service office designated in the permit.

(4) Duration of permits. The duration of permits issued under this paragraph shall be designated on the face of the

(b)(1) Application requirements for permits for incidental taking. (i) A person wishing to get a permit for an activity prohibited by \$17.31 submits an application for activities under this paragraph.

(ii) The director shall publish notice in the FEDERAL REGISTER of each application for a permit that is made under this section. Each notice shall invite the submission from interested parties, within 30 days after the date of the notice, of written data, views, or arguments with respect to the application.

(iii) Each application must be submitted on an official application (Form 3-200) provided by the Service, and must include as an attachment, all of the following information:

(A) A complete description of the activity sought to be authorized:

(B) The common and scientific names of the species sought to be covered by the permit, as well as the number, age, and sex of such species, if known;

(C) A conservation plan that speciűes:

(1) The impact that will likely result

from such taking:
(2) What steps the applicant will take to monitor, minimize, and mitigate such impacts, the funding that will be available to implement such steps, and the procedures to be used to deal with unforeseen circumstances;

(3) What alternative actions to such taking the applicant considered and the reasons why such alternatives are not proposed to be utilized; and

(4) Such other measures that the Director may require as being necessary or appropriate for purposes of the plan.

(2) Issuance criteria. (1) Upon receiving an application completed in accordance with paragraph (b)(1) of this section. the Director will decide whether or not a permit should be issued. The Director shall consider the general issuance criteria in 13.21(b) of this subchapter, except for 13.21(b)(4), and shall issue the permit if he or she finds that:

(A) The taking will be incidental;

(B) The applicant will, to the maximum extent practicable, minimize and mitigate the impacts of such takings;

(C) The applicant will ensure that adequate funding for the conservation plan and procedures to deal with unforeseen circumstances will be provided:

(D) The taking will not appreciably reduce the likelihood of the survival and recovery of the species in the wild;

(E) The measures, if any, required under paragraph (b)(1)(iii)(D) of this section will be met: and

(F) He or she has received such other assurances as he or she may require that the plan will be implemented.

(ii) In making his or her decision, the Director shall also consider the anticipated duration and geographic scope of the applicant's plauned activities, including the amount of listed species habitat that is involved and the degree to which listed species and their habitats are affected.

(3) Permit conditions. In addition to the general conditions set forth in part 13 of this subchapter, every permit issued under this paragraph shall contain such terms and conditions as the Director desms necessary or appropriate to carry out the purposes of the permit and the conservation plan including, but not limited to, monitoring and reporting requirements deemed necessary for determining whether such terms and conditions are being complied with. The Director shall rely upon existing reporting requirements to the maximum extent practicable.

(4) Duration of permits. The duration of permits issued under this paragraph shall be sufficient to provide adequate assurances to the permittee to commit funding necessary for the activities authorized by the permit, including conservation activities and land use restrictions. In determining the duration of a permit, the Director shall consider the duration of the planned activities, as well as the possible positive and negative effects associated with permits of the proposed duration on listed species. including the extent to which the conservation plan will enhance the habitat of listed species and increase the longterm survivability of such species.

§ 17.32

(5) Assurances provided to permittee in case of changed or unforeseen circumstances. The assurances in this paragraph (h)(5) apply only to incidental take permits issued in accordance with paragraph (b)(2) of this section where the conservation plan is being properly implemented, and apply only with respect to species adequately covered by the conservation plan. These assurances cannot be provided to Federal agencies. This rule does not apply to incidental take permits issued prior to March 25, 1998. The assurances provided in incidental take permits issued prior to March 25, 1998 remain in effect, and those permits will not be revised as a result of this rulemaking.

(1) Changed circumstances provided for in the plan. If additional conservation and mitigation measures are deemed necessary to respond to changed circumstances and were provided for in the plan's operating conservation program, the permittee will implement the measures specified in the plan.

(ii) Changed circumstances not provided for in the plan. If additional conservation and mitigation measures are deemed necessary to respond to changed circumstances and such measures were not provided for in the plan's operating conservation program, the Director will not require any conservation and mitigation measures in addition to those provided for in the plan without the consent of the permittee, provided the plan is being properly implemented.

(iii) Unforeseen circumstances. (A) In negotiating unforeseen circumstances, the Director will not require the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or other natural resources beyond the level otherwise agreed upon for the specles covered by the conservation plan without the consent of the permittee.

(B) If additional conservation and mitigation measures are deemed necessary to respond to unforeseen circumstances, the Director may require additional measures of the permittee where the conservation plan is being properly implemented, but only if such measures are limited to modifications within conserved habitat areas, if any, or to the conservation plan's operating

conservation program for the affected species, and maintain the original terms of the conservation plan to the maximum extent possible. Additional conservation and mitigation measures will not involve the commitment of additional land, water or financial compensation or additional restrictions on the use of land, water, or other natural resources otherwise available for development or use under the original terms of the conservation plan without the consent of the permittee.

(C) The Director will have the burden of demonstrating that such unforeseen circumstances exist, using the best scientific and commercial data available. These findings must be clearly documented and based upon reliable technical information regarding the status and habitat requirements of the affected species. The Director will consider, but not be limited to, the following factors:

(I) Size of the current range of the affected species;

(2) Percentage of range adversely affected by the conservation plan:

(3) Percentage of range conserved by the conservation plan;

(4) Ecological significance of that portion of the range affected by the conservation plan;

(5) Level of knowledge about the affected species and the degree of specificity of the species' conservation program under the conservation plan; and

- (6) Whether failure to adopt additional conservation measures would appreciably reduce the likelihood of survival and recovery of the affected species in the wild.
- (6) Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local, or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included in a conservation plan.

(7) Discontinuance of permit activity. Notwithstanding the provisions of §13.26 of this subchapter, a permittee under this paragraph (b) remains responsible for any outstanding minimization and mitigation measures reguired under the terms of the permit for take that occurs prior to surrender of the permit and such minimization

§ 17.32

and mitigation measures as may be required pursuant to the termination provisions of an implementing agreement, habitat conservation plan, or permit even after surrendering the permit to the Service pursuant to §13.26 of this subchapter. The permit shall be deemed canceled only upon a determination by the Service that such minimization and mitigation measures have been implemented. Upon surrender of the permit, no further take shall be authorized under the terms of the surrendered permit.

(8) Criteria for revocation. A permit issued under paragraph (b) of this section may not be revoked for any reason except those set forth in §13.28(a)(1) through (4) of this subchapter or unless continuation of the permitted activity would be inconsistent with the criterion set forth in 16 U.S.C. 1539(a)(2)(B)(iv) and the inconsistency has not been remedied.

(c)(1) Application requirements for permits for the enhancement of survival through Safe Harbor Agreements. The applicant must submit an application for a permit under this paragraph (c) to the appropriate Regional Director, U.S. Fish and Wildlife Service, for the Region where the applicant resides or where the proposed action is to occur (for appropriate addresses, see 50 CFR 10.22), if the applicant wishes to engage in any activity prohibited by §17.31. The applicant must submit an official Service application form (3-200.54) that includes the following information:

(i) The common and scientific names of the listed species for which the applicant requests incidental take authorization;

(ii) A description of how incidental take of the covered species pursuant to the Safe Harbor Agreement is likely to occur, both as a result of management activities and as a result of the return to baseline:

(iii) A Safe Harbor Agreement that complies with the requirements of the Safe Harbor policy available from the Service: and

(iv) The Director must publish notice in the FEDERAL REGISTER of each application for a permit that is made under this paragraph (c). Buch notice must invite the submission from interested parties within 30 days after the date of

the notice of written data, views, or arguments with respect to the application. The procedures included in §17.22(e) for permit objection apply to any notice published by the Director under this paragraph (c).

(2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (c)(1) of this section, the Director will decide whether or not to issue a permit. The Director shall consider the general issuance criteria in \$13.21(b) of this subchapter, except for §13.21(b)(4), and may issue the permit if he or she finds:

(i) The take will be incidental to an otherwise lawful activity and will be in accordance with the terms of the Safe

Harbor Agreement;

(ii) The implementation of the terms of the Safe Harbor Agreement is reasonably expected to provide a net conservation benefit to the affected listed species by contributing to the recovery of listed species included in the permit, and the Safe Harbor Agreement otherwise complies with the Safe Harbor policy available from the Service;

(iii) The probable direct and indirect effects of any authorized take will not appreciably reduce the likelihood of survival and recovery in the wild of

any listed species;

(iv) Implementation of the terms of the Safe Harbor Agreement is consistent with applicable Federal, State, and Tribal laws and regulations;

(v) Implementation of the terms of the Safe Harbor Agreement will not be in conflict with any ongoing conservation or recovery programs for listed species covered by the permit; and

(vi) The applicant has shown capability for and commitment to implementing all of the terms of the Safe

Harbor Agreement.

- (3) Permit conditions. In addition to any applicable general permit condi-tions set forth in part 13 of this subchapter, every permit issued under this paragraph (c) is subject to the following special conditions:
- (i) A requirement for the participating property owner to notify the Service of any transfer of lands subject to a Safe Harbor Agreement;

(ii) When appropriate, a requirement for the permittee to give the Service reasonable advance notice (generally

§ 17.32

at least 30 days) of when he or she exthe permitted activity would either appects to incidentally take any listed

species covered under the permit. Such notification will provide the Service with an opportunity to relocate affected individuals of the species, if possible and appropriate; and

(iii) Any additional requirements or conditions the Director deems necessary or appropriate to carry out the purposes of the permit and the Safe Harbor Agreement.

(1) Permit effective date. Permits issued under this paragraph (c) become effective the day of issuance for species covered by the Safe Harbor Agreement.

(b) Assurances provided to permittee. (1) The assurances in subparagraph (ii) of this paragraph (c)(5) apply only to Safe Harbor permits issued in accordance with paragraph (c)(2) of this section where the Safe Harbor Agreement is being properly implemented, and apply only with respect to species covered by the Agreement and permit. These assurances cannot be provided to Federal agencies. The assurances provided in this section apply only to Safe Harbor permits issued after July 19, 1999.

(ii) The Director and the permittee may agree to revise or modify the management measures set forth in a Safe Harbor Agreement if the Director determines that such revisions or modifications do not change the Director's prior determination that the Safe Harbor Agreement is reasonably expected to provide a net conservation benefit to the listed species. However, the Director may not require additional or different management activities to be undertaken by a permittee without the consent of the permittee.

(6) Additional actions. Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included in a Safe Harbor Agreement.

(7) Criteria for revocation. The Director may not revoke a permit issued under paragraph (c) of this section except as provided in this paragraph. The Director may revoke a permit for any reason set forth in §13.28(a)(1) through (4) of this subchapter. The Director may revoke a permit if continuation of

preciably reduce the likelihood of survival and recovery in the wild of any listed species or directly or indirectly alter designated critical habitat such that it appreciably diminishes the value of that critical habitat for both the sarvival and recovery of a listed species. Before revoking a permit for either of the latter two reasons, the Director, with the comment of the permittee, will pursue all appropriate options to avoid permit revocation. These

50 CFR Ch. I (10-1-08 Edition)

options may include, but are not limited to: extending or modifying the existing permit, capturing and relocating the species, compensating the landowner to forgo the activity, purchasing an easement or fee simple interest in the property, or arranging for a third-

party acquisition of an interest in the property.

(8) Duration of permits. The duration of permits issued under this paragraph (c) must be sufficient to provide a net conservation benefit to species covered in the enhancement of survival permit. In determining the duration of a permit, the Director will consider the duration of the planned activities, as well as the positive and negative effects asseciated with permits of the proposed duration on covered species, including the extent to which the conservation activities included in the Safe Harbor Agreement will enhance the survival and contribute to the recovery of listed species included in the permit.

(d)(1) Application requirements for permits for the enhancement of survival through Candidate Conservation Agreements with Assurances. The applicant must submit an application for a permit under this paragraph (d) to the appropriate Regional Director, U.S. Fish and Wildlife Service, for the Region where the applicant resides or where the proposed activity is to occur (for apprepriate addresses, see 50 CFR 10.22). When a species covered by a Candidate Conservation Agreement with Assurances is listed as threatened and the applicant wishes to engage in activities identified in the Agreement and otherwise prohibited by §17.31, the applicant must apply for an enhancement of survival permit for species covered by the Agreement. The permit will become valid if and when covered

will not be in conflict with any ongoing conservation programs for species covered by the permit; and

§ 17.32

proposed, candidate or other unlisted species is listed as a threatened apecies. The applicant must submit an official Service application form (3-200.54) that includes the following information:

(1) The common and scientific names of the species for which the applicant requests incidental take authorization;

(ii) A description of the land use or water management activity for which the applicant requests incidental take authorization; and

(iii) A Candidate Conservation Agreement that complies with the requirements of the Candidate Conservation Agreement with Assurances policy available from the Service.

(iv) The Director must publish notice in the FEDERAL REGISTER of each application for a permit that is made under this paragraph (d). Each notice must invite the submission from interested parties within 30 days after the date of the notice of written data, views, or arguments with respect to the application. The procedures included in §17.22(e) for permit objection apply to any notice published by the Director under this paragraph (d).

(2) Issuance criteria. Upon receiving an application completed in accordance with paragraph (d)(1) of this section, the Director will decide whether or not to issue a permit. The Director shall consider the general issuance criteria in §13.21(b) of this subchapter, except for §13.21(b)(4), and may issue the permit if he or she finds:

(i) The take will be incidental to an otherwise lawful activity and will be in accordance with the terms of the Candidate Conservation Agreement;

(ii) The Candidate Conservation Agreement complies with the requirements of the Candidate Conservation Agreement with Assurances policy available from the Service;

(iii) The probable direct and indirect effects of any authorized take will not appreciably reduce the likelihood of survival and recovery in the wild of any species;

(iv) Implementation of the terms of the Candidate Conservation Agreement is consistent with applicable Federal, State, and Tribal laws and regulations;

(v) Implementation of the terms of the Candidate Conservation Agreement

(vi) The applicant has shown capability for and commitment to implementing all of the terms of the Candidate Conservation Agreement.

(3) Permit conditions. In addition to any applicable general permit conditions set forth in part 13 of this subchapter, every permit issued under this paragraph (d) is subject to the following special conditions:

(i) A requirement for the property owner to notify the Service of any transfer of lands subject to a Candidate

Conservation Agreement:

(ii) When appropriate, a requirement for the permittee to give the Service reasonable advance notice (generally at least 30 days) of when he or she expects to incidentally take any listed species covered under the permit. Such notification will provide the Service with an opportunity to relocate affected individuals of the species, if posaible and appropriate; and

(iii) Any additional requirements or conditions the Director deems necessary or appropriate to carry out the purposes of the permit and the Candidate Conservation Agreement.

(4) Permit effective date. Permits issued under this paragraph (d) become effective for a species covered by a Candidate Conservation Agreement on the effective date of a final rule that lists a covered species as threatened.

(5) Assurances provided to permittee in case of changed or unforeseen circumstances. The assurances in this paragraph (d)(5) apply only to permits issued in accordance with paragraph (d)(2) where the Candidate Conservation with Assurances Agreement is being properly implemented, and apply only with respect to species adequately covered by the Candidate Conservation with Assurances Agreement. These assurances cannot be provided to Federal agencies.

(1) Changed circumstances provided for in the Agreement. If the Director determines that additional conservation measures are necessary to respond to changed circumstances and these measures were set forth in the Agreement, the permittee will implement the measures specified in the Agreement.

§17.32

- (ii) Changed circumstances not provided for in the Agreement. If the Director determines that additional conservation measures not provided for in the Agreement are necessary to respond to changed circumstances, the Director will not require any conservation measures in addition to those provided for in the Agreement without the consent of the permittee, provided the Agreement is being properly implemented.
- (iii) Unforeseen circumstances. (A) In negotiating unforeseen circumstances, the Director will not require the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or other natural resources beyond the level otherwise agreed upon for the species covered by the Agreement without the consent of the permittee.
- (B) If the Director determines additional conservation measures are necessary to respond to unforeseen circumstances, the Director may require additional measures of the permittee where the Agreement is being properly implemented, but only if such measures maintain the original terms of the Agreement to the maximum extent Additional conservation possible. measures will not involve the commitment of additional land, water, or financial compensation or additional restrictions on the use of land, water, or other natural resources otherwise available for development or use under the original terms of the Agreement without the consent of the permittee.
- (C) The Director will have the burden of demonstrating that unforeseen circumstances exist, using the best scientific and commercial data available. These findings must be clearly documented and based upon reliable technical information regarding the status and habitat requirements of the affected species. The Director whi consider, but not be limited to, the following factors:
- (1) Size of the current range of the affected species;
- (2) Percentage of range adversely affected by the Agreement;
- (3) Percentage of range conserved by the Agreement;

50 CFR Ch. I (10-1-08 Edition)

- (4) Ecological significance of that portion of the range affected by the Agreement;
- (5) Level of knowledge about the affected species and the degree of specificity of the species' conservation program under the Agreement; and
- (6) Whether failure to adopt additional conservation measures would appreciably reduce the likelihood of survival and recovery of the affected species in the wild.
- (6) Additional actions. Nothing in this rule will be construed to limit or constrain the Director, any Federal, State, local or Tribal government agency, or a private entity, from taking additional actions at its own expense to protect or conserve a species included in a Candidate Conservation with Assurances Agreement.
- (7) Criteria for revocation. The Director may not revoke a permit issued under paragraph (d) of this section except as provided in this paragraph. The Director may revoke a permit for any reason set forth in §13.28(a)(1) through (4) of this subchapter. The Director may revoke a permit if continuation of the permitted activity would either appreciably reduce the likelihood of survival and recovery in the wild of any listed species or directly or indirectly alter designated critical habitat such that it appreciably diminishes the value of that critical habitat for both the survival and recovery of a listed species. Before revoking a permit for either of the latter two reasons, the Director, with the consent of the permittee, will pursue all appropriate options to avoid permit revocation. These options may include, but are not limited to: extending or modifying the existing permit, capturing and relocating the species, compensating the landowner to forgo the activity, purchasing an easement or fee simple interest in the property, or arranging for a third-party acquisition of an interest in the property.
- (8) Duration of the Candidate Conservation Agreement. The duration of a Candidate Conservation Agreement covered by a permit issued under this paragraph (d) must be sufficient to enable the Director to determine that the benefits of the conservation measures in the Agreement, when combined with

those benefits that would be achieved if it is assumed that the conservation measures would also be implemented on other necessary properties, would preclude or remove any need to list the species covered by the Agreement.

[50 FR 39689, Sept. 30, 1985, as amended at 63 FR 8871, Feb. 23, 1998; 63 FR 52635, Oot. 1, 1998; 64 FR 32714, June 17, 1999; 64 FR 52576, Sept. 30, 1999; 69 FR 24093, May 3, 2004; 89 FR 29670, May 25, 2004; 69 FR 71731, Dec. 10, 20011

§ 17.40 Special rules—mammals.

- (a) [Reserved]
- (b) Grizzly bear (Ursus arcies)—(1) Prohibitions. The following prohibitions apply to the grizzly bear:
- (1) Taking. (A) Except as provided in paragraphs (b)(1)(i)(B) through (F) of this section, no person shall take any grizzly bear in the 48 conterminous states of the United States.
- (B) Grizzly bears may be taken in self-defense or in defense of others, but such taking shall be reported, within 5 days of occurrence, to the Assistant Regional Director, Division of Law Enforcement, U.S. Fish and Wildlife Service, P.O. Box 25486, Denver Federal Center, Denver, Colorado 80225 (303/236-7540 or FTS 776-7540), if occurring in Montana or Wyoming, or to the Assistant Regional Director, Division of Law Enforcement, U.S. Fish and Wildlife Service, Lloyd 500 Building, Suite 1490, 500 Northeast Multnomah Street, Portland, Oregon 97232 (503/231-6125 or FTS 429-6125), if occurring in Idaho or Washington, and to appropriate State and Indian Reservation Tribal anthorities. Grizzly bears or their parts taken in self-defense or in defense of others shall not be possessed, delivered, carried, transported, shipped, exported, received, or sold, except by Federal. State, or Tribal authorities.
- (C) Removal of nuisance bears. A grizzly hear consituting a demonstrable but non immediate threat to human safety or committing significant depredations to lawfully present livestock, crops, or beehives may be taken, but only if:
- (1) It has not been reasonably possible to aliminate such threat or depredation by live-capturing and releasing unharmed in a remote area the grizzly bear involved; and

(2) The taking is done in a humane manner by authorized Federal, State, or Tribal authorities, and in accordance with current interagency guidelines covering the taking of such nuisance bears; and

\$17.40

- (3) The taking is reported within 5 days of occurrence to the appropriate Assistant Regional Director, Division of Law Enforcement, U.S. Fish and Wildlife Service, as indicated in paragraph (b)(1)(i)(B) of this section, and to appropriate State and Tribal authorities.
- (D) Federal, State, or Tribal scientific or research activities. Federal, State, or Tribal authorities may take grizzly bears for scientific or research purposes, but only if such taking does not result in death or permanent injury to the bears involved. Such taking must be reported within 5 days of occurrence to the appropriate Assistant Regional Director, Division of Law Enforcement, U.S. Fish and Wildlife Service, as indicated in paragraph (b)(1)(i)(B) of this section, and to appropriate State and Tribal authorities.
 - (E) [Reserved]
- (F) National Parks. The regulations of the National Park Service shall govern all taking of grizzly bears in National Parks.
- (ii) Unlawfully taken grizzly bears. (A) Except as provided in paragraphs (b)(1Xii)(B) and (iv) of this section, no person shall possess, deliver, carry, transport, ship, export, receive, or sell any unlawfully taken grizzly bear. Any unlawfully taking of a grizzly bear shall be reported within 5 days of occurrence to the appropriate Assistant Regional Director, Division of Law Enforcement, U.S. Fish and Wildlife Service, as indicated in paragraph (b)(1)(1)(B) of this section, and to appropriate State and Tribal authorities.
- (B) Authorized Federal, State, or Tribal employees, when acting in the course of their official duties, may, for scientific or research purposes, possess, deliver, carry, transport, ship, export, or receive unlawfully taken grizzly bears.
- (iii) Import or export. Except as provided in paragraphs (b)(1)(iii) (A) and (B) and (iv) of this section, no person shall import any grizzly bear into the United States.

13.25 Transfer of permits and scope of permit authorization.

13.28 Discontinuance of permit activity.

13.27 Permit suspension.

13.28 Permit revocation.

13.29 Review procedures.

Subpart D—Conditions

13.41 Humans conditions.

13.42 Permits are specific.

13.43 Alteration of permits.

13.44 Display of permit.

13.45 Filing of reports.

13.46 Maintenance of records.

13.47 Inspection requirement.

13.48 Compliance with conditions of permit.

13.49 Surrender of permit.

13.50 Acceptance of liability.

AUTHORITY: 16 U.S.C. 568a, 704, 712, 742j-1, 1374(g), 1382, 1538(d), 1539, 1540(f), 3374, 4901-4916; 18 U.S.C. 42; 19 U.S.C. 1202; 31 U.S.C. 9701

SOURCE: 39 FR 1161, Jan. 4, 1974, unless otherwise noted.

Subpart A-Introduction

§ 13.1 General.

(a) A person must obtain a valid permit before commencing an activity for which a permit is required by this subchapter, except as provided in § 23.53 of this subchapter for retrospective permits for certain CITES shipments under very specific situations.

(b) A person must apply for such a permit under the general permit procedures of this part and any other regulations in this subchapter that apply to the proposed activity.

(1) The requirements of all applicable parts of this subchapter must be met.

(2) A person may submit one application that includes the information required in each part of this subchapter. and a single permit will be issued if appropriate.

[72 FB 46446, Aug. 23, 2007]

§ 13.2 Purpose of regulations.

The regulations contained in this part provide uniform rules, conditions. and procedures for the application for and the Issuance, denial, suspension, revocation, and general administration of all permits issued pursuant to this subchapter B.

[54 FR 38147, Sept. 14, 1989]

§ 13.5

§13.3 Scope of regulations.

The provisions in this part are in addition to, and are not in Heu of, other permit regulations of this subchapter and apply to all permits issued thereunder, including "Importation, Exportation and Transportation of Wildlife" (part 14), "Wild Bird Conservation Act" (part 15), "Injurious Wildlife" (part 16), "Endangered and Threatened Wildlife and Plants" (part 17), "Marine Mammals" (part 18), "Migratory Bird Permits" (part 21), "Eagle Permits" (part 22), and "Endangered Species Convention" (the Convention on International Trade in Endangered Species of Wild Fauna and Flora) (part 23). As used in this part 13, the term "permit" will refer to a license, permit, certificate, letter of authorization, or other document as the context may require, and to all such documents issued by the Service or other authorized U.S. or foreign government agencies.

[70 FR 18317, Apr. 11, 2005]

§ 13.4 Emergency variation from requirements.

The Director may approve variations from the requirements of this part when he finds that an emergency exists and that the proposed variations will not hinder effective administration of this subchapter B, and will not be unlawful.

§ 13.5 Information collection require-

(a) The Office of Management and Budget approved the information collection requirements contained in this part 13 under 44 U.S.C. and assigned OMB Control Number 1018-0092. The Service may not conduct or sponsor, and you are not required to respond, to a collection of information unless it displays a currently valid OMB control number. We are collecting this information to provide information necessary to evaluate permit applications. We will use this information to review permit applications and make decisions, according to criteria established in various Federal wildlife conservation statutes and regulations, on the issuance, suspension, revocation, or denial permits. You must respond to obtain or retain a permit.

§ 12.42

order the proceeds or any part thereof restored to the petitioner, after deducting from such proceeds the costs of seizure, storage, forfeiture and disposition, the duties, if any, accruing on the seized property, and any sum due on a lien for freight, charges, or contribution in general average, notice of which has been filed with the Solicitor according to law.

(g) If the Solicitor decides that relief should not be granted, the Solicitor shall so notify the petitioner in writing, stating in the notification the reasons for denying relief. The petitioner may then file a supplemental petition, but no supplemental petition shall be considered unless it is received within 60 days from the date of the Solicitor's notification denying the original petition.

[45 FR 17864, Mar. 19, 1980, as amended at 47 FR 66861, Dec. 21, 1982]

§12.42 Recovery of certain storage costs.

If any wildlife, plant, or evidentiary item is seized and forfeited under the Endangered Species Act, 16 U.S.C. 1531 et seq., any person whose act or omission was the basis for the seizure may be charged a reasonable fee for expenses to the United States connected with the transfer, board, handling, or storage of such property. If any fish, wildlife or plant is seized in connection with a violation of the Lacey Act Amendments of 1981, 16 U.S.C. 3371, et seq., any person convicted thereof, or assessed a civil penalty therefor, may be assessed a reasonable fee for ex penses of the United States connected with the storage, care and maintenance of such property. Within a reasonable time after forfeiture, the Service shall send to such person by registered or certified mail, return receipt requested, a bill for such fee. The bill shall contain an itemized statement of the applicable costs, together with instructions on the time and manner of payment. Payment shall be made in accordance with the bill. The recipient of any assessment of costs under this section who has an objection to the reasonableness of the costs described in the bill may, within 30 days of the date on which he received the bill, file written objections with the Regional Direc-

50 CFR Ch. (10-1-08 Edition)

tor of the Fish and Wildlife Service for the Region in which the seizure occurred. Upon receipt of the written objections, the appropriate Regional Director will promptly review them and within 30 days mail his final decision to the party who filed objections. In all cases, the Regional Director's decision shall constitute final administrative action on the matter.

[47 FR 56861, Dec. 21, 1982]

Subpart F—Return of Properly

§12.51 Return procedure.

If, at the conclusion of the appropriate proceedings, seized property is to be returned to the owner or consignee, the Solicitor or Service shall issue a letter or other document authorizing its return. This letter or other document shall be delivered personally or sent by registered or certified mail, return receipt requested, and shall identify the owner or consignee, the seized property, and, if appropriate, the ballee of the seized property. It shall also provide that upon presentation of the letter or other document and proper identification, and the signing of a receipt provided by the Service, the seized property is authorized to be released, provided it is properly marked in accordance with applicable State or Foderal requirements.

PART 13—GENERAL PERMIT **PROCEDURES**

Subpart A-Introduction

- General.
- 13.2 Purpose of regulations.
- 13.3 Scope of regulations.
- 13.4 Emergency variation from requirements.
- 13.5 Information collection requirements.

Subpart B—Application for Permits

- 13.11 Application procedures.
- 13.12 General information requirements on applications for permits.

Subpart C—Permit Administration

- 13.21 Issuance of permits.
- 13.22 Renewal of permits.
- Amendment of permits.
- Right of succession by certain persons.



U.S. Fish and Wildlife Serv., Interior

§ 13.11

(δ) You may obtain applications for bald and golden eagle permits (60 CFR 22) and migratory bird permits (50 CFR 21), except for banding and marking permits, by writing to the Migratory Bird Permit Program Office in the Region in which you reside. For mafling addresses for the Migratory Bird Regional Permit Offices, see below, or go http://permits.fws.gov/mbpermits/addresses.html. Send completed applications to the same address. The mailing addresses for the Regional Migratory Bird Permit Offices are as follows:

Region 1 (CA, HI, ID, NV, OR, WA, American Samoa, Guam, Commonwealth of the Northern Mariana Islands, and other Pacific possessions): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, 911 N.E. 11th Avenue, Portland, OR 97232-4181. Region 2 (AZ, NM, OK, TX): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, P.O. Box 709, Albuquerque, NM 87103.

Region 3 (IA, IL, IN, MN, MO, MI, OH, WI): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, One Federal Drive, Fort Snelling, MN

Region 4 (AL, AR, FL, GA, KY, LA, MS, NC, SC, TN, PR, VI, and Caribbean possessions): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, P.O. Box 49208, Atlanta, GA 30359.

Region 5 (CT, DC, DE, MA, MD, ME, NH, NJ, NY, PA, RI, VA, VT, WV): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, P.O. Box 779, Hadley, MA 01035-0779.

Region 6 (CO, KS, MT, ND, NE, SD, UT, WY): U.S. Fish and Wildlife Service, Migratory Bird Permit Office, P.O. Box 25486, DFC (60130), Denver, CO

80225-0486.

Region 7 (AK): U.S. Fish and Wildlife Service, Migratory Bird Permit Office (MS-201), 1011 E. Tudor Road, Anchorage, AK 99503.

(c) Time notice. The Service will process all applications as quickly as possible. However, we cannot guarantee final action within the time limit you request. You should ensure that applications for permits for marine mammals and/or endangered and threatened species are postmarked at least 90 calendar days prior to the requested effec-

tive date. The time we require for processing of endangered and threatened species incidental take permits will vary according to the project scope and significance of effects. Submit applications for all other permits to the issuing/reviewing office and ensure they are postmarked at least 60 calendar days prior to the requested effective date. Our processing time may be increased by the procedural requirements of the National Environmental Policy Act (NEPA), the requirement to publish a notice in the FEDERAL REG-ISTER requesting a 30-day public comment period when we receive certain types of permit applications, and/or the time required for extensive consultation within the Service, with other Federal agencies, and/or State or foreign governments. When applicable, we may require permit applicants to provide additional information on the proposal and on its environmental effects as may be necessary to satisfy the procedural requirements of NEFA.

(d) Fees. (1) Unless otherwise exempted under this paragraph (d), you must pay the required permit processing fee at the time that you apply for issuance or amendment of a permit. You must pay in U.S. dollars. If you submit a check or money order, it must be made payable to the "U.S. Fish and Wildlife Service." The Service will not refund any application fee under any circumstances if we have processed the application. However, we may return the application fee if you withdraw the application before we have significantly processed it.

(2) If regulations in this subchapter require more than one type of permit for an activity and the permits are issued by the same office, the issuing office may issue one consolidated permit authorizing the activity in accordance with §13.1. You may submit a single application in such cases, provided that the single application contains all the information required by the separate applications for each permitted activity. Where more than one permitted activity is consolidated into one permit, the issuing office will charge the highest single fee for the activity permitted.

(b) We estimate the public reporting burden for these reporting requirements to vary from 15 minutes to 4 hours per response, with an average of 0.803 hours per response, including time for reviewing instructions, gathering and maintaining data, and completing and reviewing the forms. Direct comments regarding the burden estimate or any other aspect of these reporting requirements to the Service Information Collection Control Officer, MS-222 ARLSQ, U.S. Fish and Wildlife Service, Washington, DC 20240, or the Office of Management and Budget, Paperwork Reduction Project (1018-0092), Washington, DC 20603.

[63 FR 52634, Oct. 1, 1998]

Subpart 8—Application for Permits

§13.11 Application procedures.

The Service may not issue a permit for any activity authorized by this subchapter B unless you have filed an application under the following procedures:

- (a) Forms. Applications must be submitted in writing on a Federal Fish and Wildlife License/Permit Application (Form 3-200) or as otherwise specifically directed by the Service.
- (b) Forwarding instructions. Applications for permits in the following categories should be forwarded to the issuing office indicated below.
- (1) You may obtain applications for migratory bird banding permits (50 CFR 21.22) by writing to: Rird Banding Laboratory, USGS Patuent Wildlife Research Center, 12100 Beech Forest Road, Laurel, Maryland 20708-4037. Submit completed permit applications to the same address.
- (2) You may obtain applications for designated port exception permits and import/export licenses (50 OFR 14) by writing to the Special Agent in Charge (SAC) of the Region in which you reside (see 50 CFR 2.2 or the Service Website, http://www.fws.gov, for addresses and boundaries of the Regions). Submit completed permit applications to the same addresse.
- (3) You may obtain applications for Wild Bird Conservation Act permits (50 CFR 15); injurious wildlife permits (50 CFR 16); captive-bred wildlife registra-

tions (50 CFR 17); permits authorizing import, export, or foreign commerce of endangered and threatened species, and interstate commerce of non-native endangered or threatened species (59 CFR 17); marine mammal permits (50 CFR 18); and permits and certificates for import, export, and reexport of species listed under the Convention on International Trade in Endangered Species of Wild Fauna and Flora (CTTRS) (50 CFR 23) from: U.S. Fish and Wildlife Service, Division of Management Authority, 4401 N. Fairfax Drive, Room 300, Arlington, Virginia 22203-1610. Submit completed permit applications to the same address.

(4) You may obtain Endangered Species Act permit applications (50 CFR 17) for activities involving native endangered and threatened species, including incidental take, scientific purnoses, enhancement of propagation or survival (i.e., recovery), and enhance-ment of survival by writing to the Regional Director (Attention: Endangered Species Permits) of the Region where the activity is to take place (see 59 OFR 2.2 or the Service Web site, http:// www.fws.gov, for addresses and boundaries of the Regions), Submit completed applications to the same address (the Regional office covering the area where the activity will take place). Permit applications for interstate commerce for native endangered and threatened species should be obtained by writing to the Regional Director (Attention: Endangered Species Permits) of the Region that has the lead for the particular species, rather than the Region where the activity will take place. You can obtain information on the lead Rogion via the Service's Endangered Species Program Web page (http://endangered.fws.gov/wildlife.html) by entering the common or scientific name of the listed species in the Regulatory Profile query box. Send interstate commerce permit applications for native listed species to the same Regional Office that has the lead for that species. Endangered Species Act permit applications for the import or export of native endangered and threatened species may be obtained from the Division of Management Authority in accordance with paragraph (b)(3) of this section.



U.S. Fish and Wildlife Serv., Interior

§ 13.12

Type of permit	CFR citation	Fee	Amendmen fee
CITES Master Pite (includes files for artificial propagation, biomedical, etc. and covers import, export, and re-export documents).	50 CFR 23	200	100
-Renewal of CITES Master File	50 CFR 23	100	
—Single-use permits issued on Master File	50 CFR 23	25	
CITES Aroual Program File	50 CFR 23	50	***************************************
-Single-use permits issued under Annual Program	50 CFR 23	25	
CITES replacement documents (lost, stolen, or damaged documents)	60 CFR 23	50	50
CITES Passport for Traveling Exhibitions and Pels	50 CFR 23	375	
CITES/ESA Passport for Traveling Exhibitions	50 CFR 23	a 100	
CITES Introduction from the Sea	50 CFB 23	100	50
CITES Participation in the Plant Rescue Center Program	50 CFR 23	0	ë
CITES Registration of Commercial Breeding Operations for Appendix-I wildlife	50 CFR 23	100	•
CITES Request for Approval of an Export Program for a State or Tribe (American Ginseng, Certain Puribagrens, and American Aligator).	50 CFR 23	(9)	e
Import/Export License	50 CFR 14	100	50
Designated Port Exception	50 CFR 14	100	50
Injurious Witdife Permit	50 CFR 16	100	50
Transport Authorization for Injurious Wildlife	50 CFR 18	25	
Wild Bird Conservation Act (WBCA)	/		
Personal Pet Import	50 CFR 15	50	****************
WBCA Scientific Research, Zoological Breeding or Display, Cooperative Breeding	50 CFR 15	100	50
WECA Approval of Cooperative Breeding Programs	50 CFR 15	200	100
-Renewal of a V/BCA Cooperative Breeding Program	50 CFR 15	50	
WBCA Approval of a Foreign Breeding Facility	50 CFR 15	4250	**************
Marine Mammal Protection Act	V	V	
Marine Alammal Public Display	50 CFR 18	300	150
Marine Marimal Scientific Research/Enhancement/Registered Agent or Tannery	60 CFR 18	150	75
 Renewal of Marine Marinal Scientific Research/Enhancement/Reg- istered Agent or Tennery. 	50 CFR 18	75	

(5) We will charge a fee for substantive amendments made to permits within the time period that the permit is still valid. The fee is generally half the original fee assessed at the time that the permit is processed; see paragraph (d)(4) of this section for the exact amount. Substantive amendments are those that pertain to the purpose and conditions of the permit and are not purely administrative. Administrative changes, such as updating name and address information, are required under 13.23(c), and we will not charge a fee for such amendments.

(6) Except as specifically noted in paragraph (d)(4) of this section, a pormit renewal is an issuance of a new permit, and applicants for permit renewal must pay the appropriate fee listed in paragraph (d)(4) of this section.

(e) Abandoned or incomplete applications. If we receive an incomplete or improperly executed application, or if you do not submit the proper fees, the issuing office will notify you of the deficiency. If you fail to supply the correct information to complete the application or to pay the required fees within 45 calendar days of the date of notification, we will consider the application abandoned. We will not refund any fees for an abandoned application.

[70 FR 18317, Apr. 11, 2005, as amended at 72 FR 48445, Aug. 23, 2007; 73 FR 29083, May 20, 2008; 73 FR 42281, July 21, 2008)

§ 13.12 General information requirements on applications for permits.

- (a) General information required for all applications. All applications must contain the following information:
- (1) Applicant's full name and address (street address, city, county, state, and zip code; and mailing address if different from street address); home and work telephone numbers; and, if available, a fax number and e-mail address, and:

§ 13.11

50 CFR Ch. I (10-1-06 Edition)

- (3) Circumstances under which we will not charge a permit application fee are as follows:
- (i) We will not charge a permit application fee to any Federal, tribal, State, or local government agency or to any individual or institution acting on behalf of such agency. Except as otherwise authorized or waived, if you fail to submit evidence of such status with your application, we will require the submission of all processing fees prior to the acceptance of the application for processing.
- (ii) As noted in paragraph (d)(4) of this section.
- (iii) We may waive the fee on a caseby-case basis for extraordinary extenuating circumstances provided that the issuing permit office and a Regional or Assistant Director approves the waiv-
- (4) User fees. The following table identifies specific fees for each permit application or amendment to a current permit. If no fee is identified under the Amendment Fee column, this particular permit either cannot be amended and a new application, and application fee, would need to be submitted or no fee will be charged for amending the permit (please contact the isming office for further information).

Type of permit	CFR citation	Fee	Amendmen fee
Migratory Bird Tree	ity Act		·
Migratory Bhd Impost/Export		\$75	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Migratory Bird Banding or Marking		***************************************	************
Migratory Bird Scientific Collecting		190	\$5
Migratory Bird Texidermy		100	40100000000000000000
Waterlowi Sale and Disposel		75	***********
Special Canada Goose			
Migratory Bird Special Purpose/Education		75	
Migratory Bird Special Purpose/Salvage	50 CFR 21	75	
Migratory Bird Special Purpose/Game Bird Propagation		75	*************
Migratory Bird Special Perpose/Miscellaneous		100	************
Falconry		100	41-941119-41119-41
Raptor Propagation		100	41000111200200000
Migratory Bird Rehabilitation		50	
Migratory Bird Depredation	50 CFR 21	100	5
Migratory Bird Depredation/Homeowner	50 CFA 21	50	***************************************
Bald and Golden Eagle Pr	rotection Act		
Eagle Scientific Collecting	50 CFR 22	100	
Eagle Exhibition		75	
Eagle Falcony		100	
Eagle - Native American Religion			
Eagle Depredation	50 CFR 22	100	f
Golden Eagle Nest Take	50 CFR 22	100	1
Eagle Transport—Scientific or Exhibition	50 CFR 22	75	
Eagle Transport-Native American Rollgious Perposes	50 CFR 22	ĕ	1
Eagle Take—Exempled under ESA			****************
Endangered Species Act/Cl	TES/Lacey Act	· · · · · · · · · · · · · · · · · · ·	L
ESA Recovery	50 CFR 17	100	,
ESA Interstate Commerce		100	
ESA Enhancement of Survival (Sale Harbor Agreement)		50	
ESA Enhancement of Survival (Candidate Conservation Agreement ances).	with Assur- 50 CFR 17	50	
ESA Incidental Take (Habital Conservation Plan)	50 CFR 17	109	1 1
ESA and CITES Import/Export and Foreign Commerce		100	1 7
ESA and CITES Museum Exchange		100	1 3
ESA Capave-bred Wildlife Registration		200	1 1
—Renevral of Captive-bred wildlife registration		100	l'`
CITES Import (Including trophies under ESA and IMMPA)		100	
and an indiana formancial archives and an analytical and and and an archive and archive archive and archive archive and archive and archive archive archive archive archive and archive archive and archive archiv	18, 23.		ŀ `
CITES Export		100	1 :
CINES Pre-Convention		76	
CITES Certificate of Origin	50 CFR 13	75	1 .
CITES Ro-Export		75	
CITES Personal Effects and Pet Export/Re-Export		50	I
CITES Appendix II Export (native furbearers and affigators—excluding to		100	

U.S. Fish and Wildlife Serv., Interior

§ 13.21

(39 FR 1161, Jan. 4, 1974, as amended at 42 FR 10465, Feb. 22, 1977; 42 FR 32377, June 24, 1977; 44 FR 51006, Sept. 17, 1979; 44 FR 59083, Oct. 12, 1979; 45 FR 56673, Aug. 25, 1980; 45 FR 78154, Nov. 25, 1980; 46 FR 42680, Aug. 21, 1981; 48 FR 31607, July 8, 1983; 48 FR 57300, Dec. 29, 1983; 50 FR 39687, Sept. 30, 1986; 50 FR 45408, Oct. 31, 1985; 54 FR 38147, Sept. 14, 1989; 70 FR 18319, Apr. 11, 2005; 72 FR 48446, Aug. 23, 2007; 73 FR 29083, May 20, 2008]

Subpart C—Permit Administration

§ 13.21 Issuance of permits.

- (a) No permit may be issued prior to the receipt of a written application therefor, unless a written variation from the requirements, as authorized by § 13.4, is inserted into the official file of the Bureau. An oral or written representation of an employee or agent of the United States Government, or an action of such employee or agent, shall not be construed as a permit unless it meets the requirements of a permit as defined in 50 CFR 10.12.
- (b) Upon receipt of a properly executed application for a permit, the Director shall issue the appropriate permit unless:
- (1) The applicant has been assessed a civil penalty or convicted of any criminal provision of any statute or regulation relating to the activity for which the application is filed, if such assessment or conviction evidences a lack of responsibility.

(2) The applicant has failed to disclose material information required, or has made false statements as to any material fact, in connection with his application:

(3) The applicant has failed to demonstrate a valid justification for the permit and a showing of responsibility:

(4) The authorization requested potentially threatens a wildlife or plant population, or

(5) The Director finds through further inquiry or investigation, or otherwise, that the applicant is not qualified.

(c) Disqualifying factors. Any one of the following will disqualify a person from receiving permits issued under this part.

(1) A conviction, or entry of a piea of guilty or nolo contendere, for a felony violation of the Lacey Act, the Migratory Bird Treaty Act, or the Bald and Golden Eagle Protection Act disqualifies any such person from receiving or exercising the privileges of a permit, unless such disqualification has been expressly waived by the Director in response to a written petition.

(2) The revocation of a permit for reasons found in §13.28 (a)(1) or (a)(2) disqualifies any such person from receiving or exercising the privileges of a similar permit for a period of five years from the date of the final agency decision on such revocation.

(3) The failure to pay any required fees or assessed costs and penalties, whether or not reduced to judgement disqualifies such person from receiving or exercising the privileges of a permit as long as such moneys are owed to the United States. This requirement shall not apply to any civil penalty presently subject to administrative or judicial appeal; provided that the pendency of a collection action brought by the United States or its assignees shall not constitute an appeal within the meaning of this subsection.

(4) The failure to submit timely, accurate, or valid reports as required may disqualify such person from receiving or exercising the privileges of a permit as long as the deficiency exists.

(d) Use of supplemental information. The issuing officer, in making a determination under this subsection, may use any information available that is relevant to the issue. This may include any prior conviction, or entry of a plea of guilty or nolo contenders, or assessment of civil or criminal penalty for a violation of any Federal or State law or regulation governing the permitted activity. It may also include any prior permit revocations or suspensions, or any reports of State or local officials. The issuing officer shall consider all relevant facts or information available, and may make independent inquiry or investigation to verify information or substantiate qualifications asserted by the applicant.

(e) Conditions of issuance and acceptance-(1) Conditions of issuance and acceptance. Any permit automatically incorporates within its terms the conditions and requirements of subpart D of

- (i) If the applicant resides or is located outside the United States, an address in the United States, aud, if conducting commercial activities, the name and address of his or her agent that is located in the United States; and
- (ii) If the applicant is an individual, the date of birth, social security number, if available, occupation, and any business, agency, organizational, or institutional affiliation associated with the wildlife or plants to be covered by the license or permit; or
- (iii) If the applicant is a business, corporation, public agency, or institution, the tax identification number; description of the type of business, corporation, agency, or institution; and the name and title of the person responsible for the permit (such as president, principal officer, or director);

(2) Location where the requested permitted activity is to occur or be conducted:

(3) Reference to the part(s) and section(s) of this subchapter B as listed in paragraph (b) of this section under which the application is made for a permit or permits, together with any additional justification, including supporting documentation as required by the referenced part(s) and section(s);

(4) If the requested permitted activity involves the import or re-export of wildlife or plants from or to any foreign country, and the country of origin, or the country of export or re-export restricts the taking, possession, transportation, exportation, or sale of wildlife or plants, documentation as indicated in §14.52(o) of this subchapter B:

(5) Certification in the following language:

I hereby certify that I have read and am familiar with the regulations contained in title 50, part 13, of the Code of Federal Regulations and the other applicable parts in subchapter B of chapter I of title 50, Code of Federal Regulations, and I further certify that the information submitted in this application for a permit is complete and accurate to the best of my knowledge and belief. I understand that any false statement herein may subject me to suspension or revocation of this permit and to the criminal penalties of 18 U.S.C. 1001.

(6) Desired effective date of permit except where issuance date is fixed by

the part under which the permit is issued:

(7) Date;

(8) Signature of the applicant; and

- (9) Such other information as the Director determines relevant to the processing of the application, including, but not limited to, information on the environmental effects of the activity consistent with 40 CFR 1596.5 and Departmental procedures at 516 DM 6, Appendix 1.3A.
- (b) Additional information required on permit applications. As stated in paragraph (a)(3) of this section certain additional information is required on all applications. These additional requirements may be found by referring to the section of this subchapter B cited after the type of permit for which application is being made:

Type of permit	Section
Importation at nondesignated ports:	
Scientific	14.31
Detedoration prevention	14.32
Economic herdship	14.33
Marking of package or container:	
Symbol meridag	14.83
Import/export ficense	14.93
Feather import quote: importation or entry	15.21
Injurious wildlife: importation or shipment	16.22
Endangered wildfile and plant permits:	
Similarity of appearance	17.52
Scientific, enhancement of propagation or	
survival, incidental taking for wildlife	17.22
Scientific, propagation, or survival for plants	17.62
Economic hardship for wildlife	17.23
Economic herdship for plants	17.63
Threatened wildlife and plant permits:	
Similarity of appearance	17.52
General for wikilie	17.32
American alligator-buyer or tanner	17.42(a)
General for plants	17.72
Marine mammels permits:	
Scientific research	18.31
Public display	18.31
Migratory bird permits:	
Sanding or marking	21.22
Scientific collecting	21.23
Taxidamist	21.24
Waterlow sale and disposal	21.25
Special eviculturist	21.26
Special purpose	21.27 21.28
Falconry	21.20
Paptor propagation permit	21.30
Depredation control	21.41
Eagle permits: Scientific or exhibition	22.21
	22.22
Indian religious 966	22.23
Depredation control	22.23
Falcottry purposes	22.24
Take of golden eagle nests	22.28
Eagle take—Exempted under ESA	
Endangered Species Convention permits	23.15

U.S. Fish and Wildlife Serv., Interlor

§ 13.25

approval of the location is a qualifying condition of the permit.

[64 FR 38149, Sept. 14, 1989, as amended at 64 FR 32711, June 17, 1999]

§13.24 Right of succession by certain persons.

- (a) Certain persons other than the permittee are authorized to carry on a permitted activity for the remainder of the term of a current permit, provided they comply with the provisions of paragraph (b) of this section. Such persons are the following:
- The surviving spouse, child, executor, administrator, or other legal representative of a deceased permittee; or
- (2) A receiver or trustee in bankruptcy or a court designated assignee for the benefit of creditors.
- (b) In order to qualify for the authorization provided in this section, the person or parsons desiring to continue the activity shall furnish the permit to the issuing officer for endorsement within 90 days from the date the successor begins to carry on the activity.
- (c) In the case of permits issued under \$17.22(b) through (d) or \$17.32(b) through (d) of this subchapter B, the successor's authorization under the permit is also subject to a determination by the Service that:
- The successor meets all of the qualifications under this part for holding a permit;
- (2) The successor has provided adequate written assurances that it will provide sufficient funding for the conservation plan or Agreement and will implement the relevant terms and conditions of the permit, including any outstanding minimization and mitigation requirements; and
- (3) The successor has provided such other information as the Service determines is relevant to the processing of the request.

[64 FR 32711, June 17, 1999]

§13.25 Transfer of permits and scope of permit authorization.

- (a) Except as otherwise provided for in this section, permits issued under this part are not transferable or assignable.
- (b) Permits issued under §17.22(b) or §17.32(b) of this subchapter B may be

transferred in whole or in part through a joint submission by the permittee and the proposed transferre or in the case of a deceased permittee, the decoased permittee's legal representative and the proposed transferre, provided the Service determines that:

- The proposed transferee meets all of the qualifications under this part for holding a permit;
- (2) The proposed transferee has provided adequate written assurances that it will provide sufficient funding for the conservation plan or Agreement and will implement the relevant terms and conditions of the permit, including any outstanding minimization and mitigation requirements; and
- (3) The proposed transferee has provided such other information as the Service determines is relevant to the processing of the submission.
- (c) In the case of the transfer of lands subject to an agreement and permit issued under \$17.22(c) or (d) or \$17.32 (c) or (d) of this subchapter B, the Service will transfer the permit to the new owner if the new owner agrees in writing to become a party to the original agreement and permit.
- (d) Except as otherwise stated on the face of the permit, any person who is under the direct control of the permittee, or who is employed by or under contract to the permittee for purposes authorized by the permit, may carry out the activity authorized by the permit.
- (e) In the case of permits issued under §17.22(b)-(d) or §17.32(b)-(d) of this subchapter to a State or local governmental entity, a person is under the direct control of the permittee where:
- The person is under the jurisdiction of the permittee and the permit provides that such person(s) may carry out the authorized activity; or
- (2) The person has been issued a permit by the governmental entity or has executed a written instrument with the governmental entity, pursuant to the terms of the implementing agreement.
- [61 FR 82711, June 17, 1999, as amended at 61 FR 62676, Sept. 30, 1999; 69 FR 24092, May 3, 2004]

§ 13.22

this part and of any part(s) or section(s) specifically authorizing or governing the activity for which the permit is issued, as well as any other conditions deemed appropriate and included on the face of the permit at the discretion of the Director.

- (2) Any person accepting and holding a permit under this subchapter B acknowledges the necessity for close regulation and monitoring of the permitted activity by the Government. By accepting such permit, the permittee consents to and shall allow entry by agents or employees of the Service upon premises where the permitted activity is conducted at any reasonable hour. Service agents or employees may enter such premises to inspect the location; any books, records, or permits required to be kept by this subchapter B; and any wildlife or plants kept under authority of the permit.
- (f) Term of permit. Unless otherwise modified, a permit is valid during the period specified on the face of the permit. Such period shall include the effective date and the date of expiration.
- (g) Denial. The issuing officer may deny a permit to any applicant who fails to meet the issuance criteria set forth in this section or in the part(s) or section(s) specifically governing the activity for which the permit is requested.

(39 FR 1161, Jan. 4, 1974, as amended at 42 FR 32377, June 24, 1977; 47 FR 30785, July 15, 1982; 54 FR 38148, Sept. 14, 1989; 70 FR 16319, Apr. 11, 2005]

§19.22 Renewal of permits.

- (a) Application for renewal. Applicants for renewal of a permit must submit a written application at least 30 days prior to the expiration date of the permit. Applicants must certify in the form required by \$13.12(a)(5) that all statements and information in the original application remain current and corrects, unless previously changed or corrected. If such information is no longer current or correct, the applicant must provide corrected information.
- (b) Renewal criteria. The Service shall issue a renewal of a permit if the applicant meets the criteria for issuance in \$13.21(b) and is not disqualified under \$13.21(c).

50 CFR Ch. I (10-1-08 Edition)

- (c) Continuation of permitted activity. Any person holding a valid, renewable permit may continue the activities authorized by the expired permit until the Service acts on the application for renewal if all of the following conditions are met:
- (1) The permit is currently in force and not suspended or revoked;
- (2) The person has complied with this section; and
- (3) The permit is not a CITES document that was issued under part 23 of this subchapter (because the CITES document is void upon expiration).
- (d) Denial. The issuing officer may deny renewal of a permit to any applicant who falls to meet the issuance criteria set forth in §13.21 of this part, or in the part(s) or section(s) specifically governing the activity for which the renewal is requested.

[54 FR 38148, Sept. 14, 1989, as amended at 72 FR 48446, Aug. 23, 2007]

§18.23 Amendment of permits.

- (a) Permittee's request. Where circumstances have changed so that a permittee desires to have any condition of his permit modified, such permittee must submit a full written justification and supporting information in conformity with this part and the part under which the permit was issued.
- (b) The Service reserves the right to amend any permit for just cause at any time during its term, upon written finding of necessity, provided that any such amendment of a permit issued under \$17.22(b) through (d) or \$17.32(b) through (d) of this subchapter shall be consistent with the requirements of \$17.22(b)(5), (c)(5) and (d)(5) or \$17.32(b)(5), (e)(6) and (d)(5) of this subchapter, respectively.
- (c) Change of name or address. A permittee is not required to obtain a new permit if there is a change in the legal individual or business name, or in the mailing address of the permittee. A permittee is required to notify the issuing office within 10 calendar days of such change. This provision does not authorize any change in location of the conduct of the permitted activity when

right to object to the proposed revocation. The issuing officer may amend any notice of revocation at any time.

- (2) Upon receipt of a notice of proposed revocation the permittee may file a written objection to the proposed action. Such objection must be in writing, must be filed within 45 calendar days of the date of the notice of proposal, must state the reasons why the permittee objects to the proposed revocation, and may include supporting documentation.
- (3) A decision on the revocation shall be made within 45 days after the end of the objection period. The issuing officer shall notify the permittee in writing of the Service's decision and the reasons therefore, together with the information concerning the right to request and the procedures for requesting reconsideration.
- (4) Unless a permittee files a timely request for reconsideration, any wildlife held under authority of a permit that is revoked must be disposed of in accordance with instructions of the issuing officer. If a permittee files a timely request for reconsideration of a proposed revocation, such permittee may retain possession of any wildlife held under authority of the permit until final disposition of the appeal process.

(54 FR 38145, Sept. 14, 1989, as amended at 64 FR 32711, June 17, 1999]

§ 13.29 Review procedures.

- (a) Request for reconsideration. Any person may request reconsideration of an action under this part if that person is one of the following:
- (1) An applicant for a permit who has received written notice of denial:
- (2) An applicant for renewal who has received written notice that a renewal is denied;
- (3) A permittee who has a permit amended, suspended, or revoked, except for those actions which are required by changes in statutes or regulations, or are emergency changes of limited applicability for which an expiration date is set within 90 days of the permit change; or
- (4) A permittee who has a permit issued or renewed but has not been granted authority by the permit to perform all activities requested in the ap-

plication, except when the activity requested is one for which there is no lawful authority to issue a permit.

- (b) Method of requesting reconsideration. Any person requesting reconsideration of an action under this part must comply with the following criteria:
- (1) Any request for reconsideration must be in writing, signed by the person requesting reconsideration or by the legal representative of that person, and must be submitted to the issuing officer.
- (2) The request for reconsideration must be received by the issuing officer within 45 calendar days of the date of notification of the decision for which reconsideration is being requested.
- (3) The request for reconsideration shall state the decision for which reconsideration is being requested and shall state the reason(s) for the reconsideration, including presenting any new information or facts pertinent to the issue(s) raised by the request for reconsideration.
- (4) The request for reconsideration shall contain a certification in substantially the same form as that provided by §13.12(a)(5). If a request for reconsideration does not contain such certification, but is otherwise timely and appropriate, it shall be held and the person submitting the request shall be given written notice of the need to submit the certification within 15 calendar days. Failure to submit certification shall result in the request being rejected as insufficient in form and content.
- (c) Inquiry by the Service. The Service may institute a separate inquiry into the matter under consideration.
- (d) Determination of grant or denial of a request for reconsideration. The issuing officer shall notify the permittee of the Service's decision within 45 days of the receipt of the request for reconsideration. This notification shall be in writing, shall state the reasons for the decision, and shall contain a description of the evidence which was relied upon by the issuing officer. The notification shall also provide information concerning the right to appeal, the official to whom an appeal may be addressed, and the procedures for making an appeal.

§ 13.26

§ 13.26 Discontinuance of permit activity.

When a permittee, or any successor to a permittee as provided for by §13.24, discontinues activities authorized by a permit, the permittee shall within 30 calendar days of the discontinuance return the permit to the issuing office together with a written statement surrendering the permit for cancellation. The permit shall be deemed void and cancelled upon its receipt by the issuing office. No refund of any fees paid for issuance of the permit or for any other fees or costs associated with a permitted activity shall be made when a permit is surrendered for cancellation for any reason prior to the expiration date stated on the face of the permit.

[64 FR 38149, Sept. 14, 1989]

§ 13.27 Permit suspension.

(a) Criteria for suspension. The privileges of exercising some or all of the permit authority may be suspended at any time if the permittee is not in compliance with the conditions of the permit, or with any applicable laws or regulations governing the conduct of the permitted activity. The issuing officer may also suspend all or part of the privileges authorized by a permit if the permittee fails to pay any fees, penalties or costs owed to the Government. Such suspension shall remain in effect until the issuing officer determines that the permittee has corrected the deficiencies.

(b) Procedure for suspension. (1) When the issuing officer believes there are valid grounds for suspending a permit the permittee shall be notified in writing of the proposed suspension by certified or registered mail. This notice shall identify the permit to be suspended, the reason(s) for such suspension, the actions necessary to correct the deficiencies, and inform the permittee of the right to object to the proposed suspension. The issuing officer may amend any notice of suspension at any time.

(2) Upon receipt of a notice of proposed suspension the permittee may file a written objection to the proposed action. Such objection must be in writing, must be filed within 45 calendar

50 CFR Ch. I (10-1-08 Edition)

days of the date of the notice of proposal, must state the reasons why the permittee objects to the proposed suspension, and may include supporting documentation.

(3) A decision on the suspension shall be made within 46 days after the end of the objection period. The issuing officer shall notify the permittee in writing of the Service's decision and the reasons therefore. The issuing officer shall also provide the applicant with the information concerning the right to request reconsideration of the decision under §13.29 of this part and the procedures for requesting reconsideration.

(54 FR 38149, Sept. 14, 1989)

§13.28 Permit revocation.

- (a) Criteria for renocation. A permit may be revoked for any of the following reasons:
- (1) The permittee willfully violates any Federal or State statute or regulation, or any Indian tribal law or regulation, or any law or regulation of any foreign country, which involves a violation of the conditions of the permit or of the laws or regulations governing the permitted activity: or
- (2) The permittee fails within 60 days to correct deficiencies that were the cause of a permit suspension; or
- (3) The permittee becomes disqualified under § 13.21(c) of this part; or
- (4) A change occurs in the statute or regulation authorizing the permit that prohibits the continuation of a permit issued by the Service; or
- (5) Except for permits issued under \$17.22(b) through (d) or \$17.32(b) through (d) or \$17.32(b) through (d) of this subchapter, the population(s) of the wildlife or plant that is the subject of the permit declines to the extent that continuation of the permitted activity would be detrimental to maintenance or recovery of the affected population.
- (b) Procedure for revocation. (1) When the issuing officer believes there are valid grounds for revoking a permit, the permittee shall be notified in writing of the proposed revocation by certified or registered mail. This notice shall identify the permit to be revoked, the reason(s) for such revocation, the proposed disposition of the wildlife, if any, and inform the permittee of the

Pt. 14

U.S. Fish and Wildlife Serv., Interior

States and permittees conducting commercial activities in the United States who reside or are located outside the United States must maintain records at a location in the United States where the records are available for inspection.

[39 FR 1161, Jan. 4, 1974, as amended at 42 FR 32377, June 24, 1977; 54 FR 38150, Sept. 14, 1989; 72 FR 48446, Aug. 23, 2007]

§13.47 Inspection requirement.

Any person holding a permit under this subchapter B shall allow the Director's agent to enter his premises at any reasonable hour to inspect any wildlife or plant held or to inspect, audit, or copy any permits, books, or records required to be kept by regulations of this subchapter B.

(39 FR 1161, Jan. 4, 1974, as amended at 42 FR 32377, June 24, 1977]

§18.48 Compliance with conditions of permit.

Any person holding a permit under subchapter B and any person acting under authority of such permit must comply with all conditions of the permit and with all applicable laws and regulations governing the permitted activity.

(54 FR 38150, Sept. 14, 1989)

§13.49 Surrender of permit.

Any person holding a permit under subchapter B shall surrender such permit to the issuing officer upon notification that the permit has been suspended or revoked by the Service, and all appeal procedures have been exhausted.

[64 FR 38160, Sept. 14, 1989]

§ 13.60 Acceptance of liability.

Except as otherwise limited in the case of permits described in §13.25(d), any person holding a permit under this subchapter B assumes all liability and responsibility for the conduct of any activity conducted under the authority of such permit.

[64 FR 32711, June 17, 1999]

PART 14—IMPORTATION, EXPORTATION, AND TRANSPORTATION OF WILDLIFE

Subpart A—Introduction

Sec.	
14.1	Purpose of regulations.
14.2	Scope of regulations.
14.3	Information collection requirements.
14.4	
gr.	

Subpart B-Importation and Exportation at Designated Ports

14.11 General restrictions.

14.12	Designated ports.
14.13	Emergency diversion.
14.14	In-transit shipments.
14.15	Personal baggage and household ef-
fe	ects.
14.16	Border ports.
14.17	Personally owned pet birds,
14.18	
14.19	Special ports.
14,20	Exceptions by permit.
14,21	Shellfish and fishery products.
14.22	Certain antique articles.
14.23	Live farm-raised fish and farm-raised
	sh eggs.
	Scientific specimens.

Subpart C-Designated Port Exception Permits

14.31 Permits to import or export wildlife at nondesignated port for scientific pur-

14.32 Permits to import or export wildlife at nondesignated port to minimize deterioration or loss.

14.33 Permits to import or export wildlife at nondesignated port to alleviate undue sconomic hardship.

Subpart D (Reserved)

Subpart E-Inspection and Clearance of Wildlife

14.51	Inspection of wildlife.
14.52	Ciearance of imported wildlife.
14.53	
14.54	Unavailability of Service officers.
14.55	Exceptions to clearance requirements
	Cultural E Mildide Declarations

Subpart F—Wildlife Declarations

Import declaration requirements. 14.62 Exceptions to import declaration requirements. 14.63 Export declaration requirements. 14.64 Exceptions to export declaration requirements.

Subpart G [Reserved]

50 CFR Ch. I (10-1-08 Edition)

§ 13.41

(e) Appeal. A person who has received an adverse decision following submission of a request for reconsideration may submit a written appeal to the Regional Director for the region in which the issuing office is located, or to the Director for offices which report directly to the Director. An appeal must be submitted within 45 days of the date of the notification of the decision on the request for reconsideration. The appeal shall state the reason(s) and issue(s) upon which the appeal is based and may contain any additional evidence or arguments to support the appeal.

(f) Decision on appeal. (1) Before a decision is made concerning the appeal the appellant may present oral arguments before the Regional Director or the Director, as appropriate, if such official judges oral arguments are necessary to clarify issues raised in the written record.

(2) The Service shall notify the appellant in writing of its decision within 45 calendar days of receipt of the appeal, unless extended for good cause and the appellant notified of the extension.

(3) The decision of the Regional Director or the Director shall constitute the final administrative decision of the Department of the Interior.

[54 FR 33149, Sept. 14, 1989]

Subpart D—Conditions

§ 13.41 Humane conditions.

Any live wildlife possessed under a permit must be maintained under humane and healthful conditions.

[54 FR 38150, Sept. 14, 1989]

§ 13.42 Permits are specific.

The authorizations on the face of a permit that set forth specific times, dates, places, methods of taking or carrying out the permitted activities, numbers and kinds of wildlife or plants, location of activity, and associated activities that must be carried out; describe certain circumscribed transactions; or otherwise allow a specifically limited matter, are to be strictly interpreted and will not be interpreted to permit similar or related matters outside the scope of strict construction.

[70 FR 18320, Apr. 11, 2905]

§ 13.43 Alteration of permits.

Permits shall not be altered, erased, or mutilated, and any permit which has been altered, erased, or mutilated shall immediately become invalid. Unless specifically permitted on the face thereof, no permit shall be copied, nor shall any copy of a permit issued pursuant to this subchapter B be displayed, offered for inspection, or otherwise used for any official purpose for which the permit was issued.

§13.44 Display of permit.

Any permit issued under this part shall be displayed for inspection upon request to the Director or his agent, or to any other person relying upon its existence.

§ 13.45 Fling of reports.

Permittees may be required to file reports of the activities conducted under the permit. Any such reports shall be filed not later than March 31 for the preceding calendar year ending December 31, or any portion thereof, during which a permit was in force, unless the regulations of this subchapter B or the provisions of the permit set forth other reporting requirements.

§ 13.46 Maintenance of records.

From the date of issuance of the permit, the permittee shall maintain complete and accurate records of any taking, possession, transportation, sale, purchase, barter, exportation, or importation of plants obtained from the wild (excluding seeds) or wildlife pursuant to such permit. Such records shall be kept current and shall include names and addresses of persons with whom any plant obtained from the wild (excluding seeds) or wildlife has been purchased, sold, bartered, or otherwise transferred, and the date of such transaction, and such other information as may be required or appropriate. Such records shall be legibly written or reproducible in English and shall be maintained for five years from the date of expiration of the permit. Permittees who reside or are located in the United